

MENTAL CAPACITY REPORT: COMPENDIUM

June 2021 | Issue 114



Welcome to the June 2021 Mental Capacity Report. Highlights this month include:

(1) In the Health, Welfare and Deprivation of Liberty Report: substance over form in DoLS authorisations, complex questions of coercion in medical treatment, and the limits of fluctuating capacity in the context of sex;

(2) In the Property and Affairs Report: a brisk dismissal of an attempt to appeal a judgment of Senior Judge Hilder about charging by a deputy, and easy read guides to making LPAs;

(3) In the Practice and Procedure Report: an important rapid consultation on hearings and the judicial view of remote hearings;

(4) In the Wider Context Report: the CPR responds to vulnerability, strengthening the right to independent living, capacity in the rear view mirror and the ECHR and the CRPD at loggerheads;

(5) In the Scotland Report: the Mental Welfare Commission on hospital discharges, change at Scottish Government (but how much) and welfare guardianships and deprivation of liberty.

You can find our past issues, our case summaries, and more on our dedicated sub-site <u>here</u>, where you can also find updated versions of both our capacity and best interests guides. We have taken a deliberate decision not to cover all the host of COVID-19 related matters that might have a tangential impact upon mental capacity in the Report. Chambers has created a dedicated COVID-19 page with resources, seminars, and more, <u>here</u>; Alex maintains a resources page for MCA and COVID-19 <u>here</u>, and Neil a page <u>here</u>. If you want more information on the Convention on the Rights of Persons with Disabilities, which we frequently refer to in this Report, we suggest you go to the <u>Small Places</u> website run by Lucy Series of Cardiff University.

Editors

Alex Ruck Keene Victoria Butler-Cole QC Neil Allen Annabel Lee Nicola Kohn Katie Scott Rachel Sullivan Stephanie David Simon Edwards (P&A)

Scottish Contributors

Adrian Ward Jill Stavert

The picture at the top, "Colourful," is by Geoffrey Files, a young man with autism. We are very grateful to him and his family for permission to use his artwork.

Contents

HEALTH, WELFARE LIBERTY			
<i>Re JB</i> : Supreme Court	exped	ites hearing	2
Capacity guide – relev	ant inf	ormation	3
The DHSC LPS factsh	eets		3
Capacity, pregnancy, r	isk and	d the courts	3
Substance or form: wh invalidate a DoLS auth		•	9
Deprivation of liberty: a (relatively) routine	•	0	10
Anorexia – why capac	ity ma	tters	14
Anorexia – the Court c			
Dialysis – how far is c	oercio	n justified?	20
Securing autonomy – balance			22
COVID-19 vaccination code for mandatory va reminder you can't fail	accinat a capa	ion (and a acity assessmer	
Sex and the limits of fl			
PROPERTY AND AFFA	AIRS		37
Short note: permissior deputy charging case.			37
Mencap publishes gui LPAs			
PRACTICE AND PROC	EDUR	Ξ	38
Consultation on hybrid	d and r	emote hearings	38
Remote hearings: the	judicia	l view	38
THE WIDER CONTEXT	-		40

CPR – Participation of Vulnerable Parties and Witnesses			
Short note: capacity in the rearview mirror40			
LeDer report published41			
Strengthening the Right to Independent Living 			
Seni's Law consultation 43			
Research corner			
Short note: voting, capacity and the ECHR on the CRPD			
Short note: the common law and the MCA – the view from Singapore			
SCOTLAND			
The hospital discharge scandal: MWC Report 47			
Change at Scottish Government – but how much?51			
Welfare guardianship and deprivation of liberty 52			
Habitual residence of a child54			
Evidence of solicitor for granter of Will54			
Standing of ECHR Article 655			
Electronic and digital Wills			

HEALTH, WELFARE AND DEPRIVATION OF LIBERTY

Re JB: Supreme Court expedites hearing

The Supreme Court has expedited to 15 July the hearing of the Official Solicitor's appeal against the decision of the Court of Appeal in <u>A Local</u> <u>Authority v JB</u> [2020] EWCA Civ 735. This will be the first time that the Supreme Court considers the vexed, and vexing question of capacity and

sexual relations.

Capacity guide - relevant information

As discussed in Alex's shedinar on <u>capacity</u> <u>fundamentals</u>, one of the most common sources of unnecessary complexity in thinking about whether a person has capacity to make a decision is not being clear about what the information is that is relevant to that decision – and what information is not relevant.

For some years, the <u>capacity guidance note</u> the 39 Essex Chambers Mental Capacity Report team produces has had an annex setting out the information that the courts have indicated is relevant (and irrelevant) for categories of decisions such as residence, care, contact, sexual relations and deprivation of liberty. We have now made it a standalone document, so it can be accessed directly. You can find it <u>here</u>.

NB, the courts have <u>emphasised</u> that the guidelines that have been set down are the starting point for considering different types of decision, as the information must always be tailored to the actual decision in question, but this will at least (1) both save you reinventing the wheel; and (2) if you start with the information as potentially relevant (or irrelevant) you will be doing so on the basis that you will be following a path adopted as appropriate by the courts.

The DHSC LPS factsheets

The DHSC published on 11 June six further factsheets about the LPS:

1. <u>Liberty Protection Safeguards: criteria for</u> <u>authorisation</u> (June 2021)

2. <u>Liberty Protection Safeguards: the appropriate</u> <u>person and Independent Mental Capacity</u> Advocates (June 2021)

3. <u>Liberty Protection Safeguards: the Approved</u> <u>Mental Capacity Professional Role</u> (June 2021)

4. <u>Liberty Protection Safeguards: Deprivation of</u> <u>liberty and authorisation of steps necessary for</u> <u>life-sustaining treatment or vital acts (Section</u> <u>4b)</u> (June 2021)

5. <u>Liberty Protection Safeguards: authorisations,</u> <u>renewals and reviews</u> (June 2021)

6. <u>Liberty Protection Safeguards: the right to</u> <u>challenge an authorisation in court</u> (June 2021)

The other factsheets can be found <u>here</u>. Alex's resources page gathering together materials and including video walkthroughs of the LPS and what can be done in pre-implementation is <u>here</u>. Neil's LPSLaw website can be found <u>here</u>.

Capacity, pregnancy, risk and the courts

A NHS Foundation Trust v An Expectant Mother [2021] EWCOP 33 (Holman J)

Best interests – birth arrangements

Summary

This case, which has attracted considerable discussion, concerned an expectant woman, aged 21 with what was described as severe agoraphobia. She had only left her house on a handful of occasions in the preceding four or five years, each time experiencing overwhelming sensations of anxiety, shortness of breath, dizziness and palpitations. She was pregnant and, in circumstances that are not set out in detail in what is a relatively short judgment, the medical teams responsible for her (it appears from the fact that there are two Trusts named in the title that one must have included a mental health team) considered that (1) the risks to her and her baby were sufficient if a home birth did not go to plan; and (2) she would not agree to leave her home to go to hospital in the event that a transfer was required. The Trusts therefore applied to court for endorsement of a plan which would see the mother transferred to hospital before she went into labour.

As to the timing of the application, Holman J referred himself to the decision of MacDonald J in <u>East Lancashire Hospitals NHS Trust v</u> <u>GH</u> [2021] EWCOP 18, which had been brought on an emergency basis whilst the woman (again, suffering from agoraphobia) was undergoing an obstructed labour, granted late at night, and where the woman had, in fact, given birth at home before the arrangements endorsed could be implemented. Holman J noted that:

15. It is, of course, possible to draw conflicting messages from that case and judgment. It could be said to illustrate that, even when doctors consider that there is an acute emergency, it may yet pass and the birth may take place at home without (apparently) long-term damage. It could be said to illustrate the capacity in some situations for a hospital to obtain an urgent hearing and an order, even in the middle of the night, when an emergency has actually arisen. But the hearing will have taken time to set up, and itself lasted nearly two hours, far too long if there had been what Professor Walker described as "a blue light ambulance emergency".

16. Overall, the case illustrates, in my view, the need to anticipate problems of this kind and to face up to them as best one can in advance, even if that involves speculation and/or reliance upon statistics. In my view, therefore, it was entirely justifiable and appropriate that the hospital trusts in the present case have made the present application. Amongst other advantages, it has enabled a thorough and informed investigation to take place, as well as a fair and transparent hearing, lasting many hours, in which to test out the issues and the evidence, all of which is impossible in any kind of emergency outof-hours situation. Further, the mother herself has been able to participate, and has participated, throughout the hearing, at a time when she is not in labour, pain or distress

As to the woman's capacity, the evidence before the court, including that of an independent expert psychiatrist, Dr Glover, was that the mother's agoraphobia was:

7. [...] so overwhelming that it exerts a significant effect on her ability to weigh matters in the balance if the activity in point entails her leaving her home. Further, in the opinion of Dr Glover, the mother has short-term memory problems which limit her capacity to manage and process complex, multifaceted information.

8. For these reasons, Dr Glover and the consultant perinatal psychiatrist for the applicant trusts both agree, as do I, that the mother lacks capacity to make decisions about whether her baby should be born at home or in hospital. Put simply, she is so overwhelmed by her agoraphobia that she is unable to weigh and process relevant considerations and unable to make any sort of decision about it. I am, accordingly, quite satisfied - and the Official Solicitor on her behalf now agrees - that the mother lacks capacity to make decisions about the location of the delivery of her baby, and also lacks litigation capacity in relation to that issue, and I will so declare.

This therefore meant that Holman J was required to consider what was in the mother's best interests. He made clear (at paragraph 11) that he did not consider that the case was about the advantages or disadvantages of hospital birth or home birth, or vice- versa, upon which capacitous women may have different views and about which a capacitous expectant mother normally has autonomous and complete freedom of choice. However, he noted that the choice was normally made in the knowledge that if, during a home birth, a medical emergency arises which may imperil the wellbeing or even the life of the mother or the baby, the mother can be fairly rapidly transferred to a hospital if required. As he noted:

The nub of this case is the potential difficulty of transferring this particular mother to hospital if a medical emergency arose, but she was so overcome by her agoraphobia that she would not go. That, of course, could potentially occur at any time of day or night, or during a weekend, when far fewer resources might be available than if the mother is taken to hospital, not in labour, in a planned way so as to give birth there.

The evidence before the court (limited somewhat by the fact that the mother had not attended hospital for scans, as a result of her agoraphobia) was that were no specific indicators that the mother would not have an uneventful spontaneous labour and vaginal delivery. Holman J, however, identified that:

13. [...] although child birth is the most natural of human events, it is not risk free. There are statistics which indicate that, in about 45 per cent of cases in which a young, healthy, primigravida mother embarks on a home birth, she will be transferred to hospital before the birth occurs. Of that 45 per cent, about one auarter are for urgent medical emergencies. The remainder are for important but less urgent reasons, such as a failure to progress, pain relief, or repairing tears. On that basis, about 10 per cent of all such home births require an urgent transfer to hospital for serious medical emergencies, threatening the mother and/or the baby. Professor James Walker, the consultant obstetrician instructed as an expert witness by the Official Solicitor, considered that what he called "urgent blue light ambulance transfers" occur during about one to two per cent of home births.

14. There is an overall statistic that about one in 200 hospital births tragically result in a still-born or otherwise seriously damaged baby. In cases which start as home births, that figure doubles to about one in 100. The difference between the two figures is largely, if not wholly, attributable to delays in effecting a transfer from the home to the hospital.

In the circumstances, and whilst "the medical witnesses do not in any way predict that there will be any emergency; [...] on the basis of those known statistics, they must, appropriately and responsibly, anticipate the possibility that there may be."

How, then, to respond to this risk? The doctors agreed, and, on the woman's behalf, the Official Solicitor agreed – an agreement endorsed by Holman J that it was: *"preferable, and in the overall*"

best interests of this particular mother and her baby, that she should give birth in hospital in a planned way around the [estimated due date], but before she goes into spontaneous labour" (paragraph 21). Having started with the medical position, Holman J also factored in the views of the woman, her partner, and her own mother:

- The woman, who participated by video, was clear that she would prefer to give birth at home, on the basis of her agoraphobia and fear of going out. Later in the judgment (at paragraph 30), he identified that there was a "known, if small, risk that, if a pre-planned birth cannot be achieved, some acute emergency may (I stress, may) arise in the home from which the mother cannot be rescued before some catastrophe occurs to either her or her baby. The risk may be low, but that which is at risk could not be potentially more grave. The mother is very, very clear that she does not want that to happen;"
- Her partner and her mother whose views fell to be considered by reference to s.4(7) MCA 2005, also were identified as preferring a hospital birth if it could be achieved.

It was also agreed that it would be in the best interests of the woman that sedation could be administered to her "so as to calm her and help her cope with the transfer." That could be administered orally or by intramuscular injection. Holman J noted that "[s]he does not like needles, but she accepts the need for injections and is not, as such, resistant to them. Indeed, moments before I commenced this judgment, she told me that the visiting midwife had just taken blood samples from her with no problem." It is important to note that no one sought to persuade Holman J (and he did not find) that the mother lacked capacity to make the decision

23. [...] The mother does have the capacity to make an informed choice between those methods, and unless is significant medical there а contraindication, the hospital will respect and be guided by her choice on the day. Similarly, if she elects a Caesarean section, she has the capacity to make an informed choice between a local or a general anaesthetic. Currently, she has clearly said that, if she is to have a Caesarean, she would prefer a general anaesthetic and to wake up when it is all over. But, of course, she has the right and capacity to change her mind about that until the last moment

whether to give birth vaginally at the hospital

(after being induced) or by Caesarean section:

The one area of disagreement between the representatives was as to the extent of additional force or restraint which could lawfully be used on a pre-planned transfer and admission, if the woman was not actually in labour and no actual acute medical emergency has actually arisen. The Trusts sought provision for the use of (the minimum necessary) reasonable force. As Holman J noted, in an emergency, "[i] the Official Solicitor has, herself, agreed and accepted that force and restraint could, if it became necessary, be used. But she submits that, short of an actual current emergency, it is not justifiable or proportionate to use force or restraint for a pre-planned admission, however desirable such an admission might otherwise be" (paragraph 26).

The medical witnesses made clear that this was

a finely-balanced decision:

28. [...] Both psychiatrists have agreed that, if force is used, that may (this again is speculative) have a damaging psychological effect on this already agoraphobic person. It may entrench her agoraphobia. It may damage or impair her bonding with her baby. It may give her flashbacks. lona-term lt mav compromise her attitude to future pregnancies, or her dealings with persons in authority. On behalf of the Official Solicitor, Ms Sophia Roper submits that these are known risks from the use of force or restraint which outweigh the more speculative and statistical risks, if the mother goes into labour at home but may then require an urgent transfer to hospital.

Having outlined what the Trusts proposed, Holman J considered that this was:

30. [...] an unattractive scenario and, on the face of it, if resorted to, a severe infringement of the mother's personal autonomy and liberty. But, on the other side of the balance here, there is the known, if small, risk that, if a pre-planned birth cannot be achieved, some acute emergency may (I stress, may) arise in the home from which the mother cannot be rescued before some catastrophe occurs to either her or her baby. The risk may be low, but that which is at risk could not be potentially more grave. The mother is very, very clear that she does not want that to happen.

31. Having very anxiously weighed and considered all the factors in this case, I am, on balance, satisfied, albeit in disagreement with the Official Solicitor, that it will be in the overall best interests of this mother if - if the necessity for it arises on the day - some trained and professional force and restraint are used to transport her to hospital, and I will so declare. The declaration will incorporate the final "care plan for delivery" of the baby, which has been amended by me and counsel during the course of the hearing. An official transcript will be made as soon as possible of this judgment, and an anonymised version of the care plan and the order will be annexed to it.

In a postscript, Holman J recorded that:

The judge was later informed that on 22 May 2021 the mother went into spontaneous labour at home. She contacted the hospital and travelled there with the support of her partner and mother and the community midwife. While still at home, she received 2mg of Lorazepam orally. Although initially resistant, she was guided by staff and her family into the ambulance and no restraint was required or used. A few hours later she was safely delivered of a healthy baby boy with a good birth weight. She returned home with the baby within the next day or two.

Comment

This case has provoked strong reactions, which have been meticulously documented by the <u>Open Justice Court of Protection project</u>, which also commented critically on the fact that Holman J held the hearing in public, but did not permit members of the public to attend remotely. It should perhaps be noted that this is likely to be an issue which is going to be encountered more frequently as the courts potentially move back to a world where at least some hearings are held in physical court rooms where no-one attends remotely (there are, of course, a whole host of – sometimes finely balanced – issues about whether remote or inperson hearings are, substantively, better for the delivery of justice for the actual parties concerned). At that point, real questions are going to arise as whether remote access for members of the public – in effect broadcasting – is or should be required. These are definitely questions for another day, but they are going to be need to be answered in due course.

Turning to the **substance** of the decision, much of the focus of the discussion has been on (in essence) the excess risk aversion of the medical professionals involved, leading to a situation where, to pre-empt a small but very serious risk, entirely disproportionate steps were sought, agreed (in part) by the very person appointed to act as her representative, and endorsed by the court. Each of these criticisms raise entirely legitimate points (and Alex and Neil have cowritten a few years ago a <u>detailed article</u> examining the procedural justice problem caused by the role that the Official Solicitor is required to play).

However, each of the criticisms need to be placed in their context by reference to some points which may risk getting otherwise lost.

The first is to note that the woman's voice – ironically – seems at some points to have been lost in some of the debates. For ourselves, and having been hoping for the end of the story to come out, we would dearly love to be able to find out from her now about (1) why she contacted the hospital herself, as the postscript described her as having done; and (2) how she felt both about the process and the outcome, in circumstances where one construction of the evidence is (from a CRPD perspective) her overriding will was to be delivered of her baby safely, with all other considerations being secondary. We would not, though, purport to speak for her.

The second is that the vast majority of birthplanning decisions relating to those with mental health conditions ¹ are undertaken in collaboration with the woman, outside the court arena. That work almost invariably includes detailed and careful advance care planning (for a specific tool for use in the context of bipolar disorder, which can be adapted to address birth plans, see the <u>PACT</u> approach).

The third is that the woman's case may raise legitimate questions about the risk analysis undertaken by the two trusts involved in the context of their planning (involving, it would appear from the judgment, not just doctors, but also the woman's community midwife and community psychiatric nurse). However, again, by way of context, it is important to recall that that approaches to risk on the behalf of medical professionals are dictated in part by anticipation of societal/regulatory responses in the event of a statistically low but substantively very high risk event coming to pass.

The fourth is that there will be some circumstances in which the interventions considered necessary to secure against the risks

¹ Whatever the position might be about agoraphobia generally, it appears clear from the judgment that the

woman's agoraphobia was sufficiently serious that she was under the care of a mental health Trust.

June 2021 Page 9

go beyond those which can be catered for under the provisions of ss.5-6 MCA 2005. At that point, it is clear that Trusts should consider approaching the court for endorsement of the plan: see, in this regard, the Serious Medical Treatment Guidance issued by Hayden J in January 2020. At one level, some of the criticisms of the judgment amount to a challenge to the idea that Trusts should contingency plan, and seek the endorsement of the court for a contingency plan. To the extent that they do, it seems that, with respect, those criticisms are misplaced, if the alternative is to leave things to unfold and then seek authority only if then required. That does not mean that, if the contingency plan is put before the court, it should not be rigorously stress-tested, but such stress-testing can in practice only take place if there is time to do so.

The fifth is that, if the Trust(s) do come to court, it is, in reality, impossible for a judge not to take into account the potential that they might endorse a plan which could lead to the death of a viable baby. Perhaps some of the discomfort expressed in relation to this case reflects the fact that Holman J was not allowed to hold that the foetus had its own independent legal rights. Had he been able to do so, he could have conducted an express balancing exercise of those rights against (if, in reality, they were against - we do not seek to give voice to the woman here) the rights of the woman. Because he could not then, in line with all other judges who have been in a similar position, he had to take them into account through an analysis of the mother's best interests which took into account her desire to be delivered safely of her baby. There are undoubtedly cases - although not necessarily this one - in which that analysis comes

uncomfortably close to a misleading legal fiction.

Substance or form: when do repeated errors invalidate a DoLS authorisation?

Re YC [2021] EWCOP 34 (Senior Judge Hilder)

Article 5 ECHR – DoLS authorisations

Summary

What happens when the paperwork authorising a deprivation of liberty under the DoLS regime repeatedly refers to the wrong person? This question was posed before Senior Judge Hilder in the context of an appeal against the dismissal of a challenge to a DoLS authorisation purportedly granted by the City of Westminster in respect of a woman, YC. The 'Form 5' document, containing – in effect – the record of authorisation, referred to the wrong name 19 out of 25 times.

On the facts of the case, in circumstance where it was accepted on behalf of YC that the **underlying** assessments had been properly conducted, Senior Judge Hilder (at paragraph 74) noted that:

The impression is [...] created that standardised phrases have been used in the administrative process of writing up a decision – which, I would suggest, is very poor practice – but overall, the frequency with which the same error appears points much more clearly to administrative, than substantive, inadequacy.

The wider headline points are that:

(1) There is a material difference between the situation where assessments have been completed correctly but there are mistakes

made in the form in which the authorisation is recorded, and where the 'outcome leads the process,' as in the <u>Neary</u> case (and also, although not brought to Senior Judge Hilder's attention, this <u>disciplinary case</u> before Social Work England where repeated plagiarisation of forms by a BIA gave rise to a risk that individuals had been – substantively – unlawfully deprived of their liberty);

- (2) Errors of form do not necessarily invalidate the authorisation. Even in the serious domain of authorisations of deprivation of liberty, there is room for a degree of pragmatic realism, as is recognised in the ECHR decisions contrasting ex facie invalid orders and prima facie valid ones, and by the 'correcting' provisions of the Mental Health Act. As Senior Judge Hilder noted at 75: "[w]here paragraph there are standardised documents, and inevitable use of information technology (including the availability of 'cut and paste'), it would be disproportionate to conclude that every error of form invalidates Form 5."
- (3) Senior Judge Hilder endorsed the following "workable and appropriate" procedure as good practice to improve the prospects of identifying and addressing errors promptly, if necessary by a completely new assessment process:
 - a. Firstly, the person granting the authorisation should carefully check that all details on Form 5 accurately reflect the other DOLS forms and relate to the particular P;
 - b. The Form 5 should be checked for accuracy by another member of the

DOLS authorisation team of the supervisory body;

- c. Form 5 should be provided to the RPR with a covering letter requesting that the RPR carefully checks that the forms, and all the information in them accurately relates to the relevant person;
- d. An express requirement for the RPR to confirm accuracy to the supervisory body would be disproportionate but the RPR could do so.

Comment

In light of this decision, supervisory bodies will no doubt be checking (1) their administrative processes to seek to eliminate cut and paste disease; and (2) their communications with RPRs.

One important point to note is that Senior Judge Hilder was emphasising that the RPR has an important practical role in making sure that the paperwork is, in fact, in order. However, she was **not** holding that this is required as a matter of law so that if the RPR misses something the authorisation itself becomes invalid. Of course, if it becomes clear that an RPR – especially a paid RPR – has simply put the authorisation in a drawer without noticing errors, especially glaring errors, that should be a red flag for whether the RPR is actually able to undertake their tasks under paragraph 140 of Schedule A1 DoLS of representing and supporting the person.

Deprivation of liberty: a spotlight on the (relatively) routine

AA v GA and An NHS Clinical Commissioning Group [2020] EWFC B67 (HHJ Pemberton)

Article 5 ECHR – DoLS authorisations

Summary

In this case, HHJ Pemberton considered a s.21A challenge to a standard authorisation of GA's deprivation of liberty at 'Placement A'. AA is GA's mother, and GA's care was provided by the CCG. GA was 22 years old, and had diagnoses of a severe learning disability, Autism Spectrum Disorder, ADHD, epilepsy and severe communication difficulties. The parties were in agreement that GA lacked capacity to make decisions as to her residence and care.

All parties appeared to be in agreement that it was in GA's best interests to move from Placement A, and the decision before the court was whether it would be in her best interests:

- 1. To move to an alternative specialist residential placement, 'Placement B'; or
- 2. To return to live with AA and GA's brother at the family home with no professional input from the CCG.

The standard authorisation was granted in January 2020 initially for a five-month period, but was extended by the court until the date of its judgment (approximately six months after the initial expiration date).

GA had lived in the family home with her mother and siblings until November 2019. The CCG had made an application in autumn 2019 to remove GA from the family home and accommodate her in a residential unit, arguing that *"significant concerns had arisen over the past 12 months that GA's needs were not being met in the family home. In addition, some child protection concerns in respect of the younger siblings had arisen which had caused children's services to become involved* with the family and for the children to be recorded as being at risk of significant harm due to physical outbursts from GA including physical assaults towards staff, her mother and siblings (I accept that GA did not intend to hurt anyone)'" (paragraph 11). The CCG had provided carers in the home, but one agency withdrew in July 2019; GA also attended day services during the week. The court approved the application to move to Placement A in December 2019, and it was planned that "a full functional, sensory and communication assessment was to be undertaken" (paragraph 15) while GA was there. This application was supported by GA's litigation friend.

In February 2020, concerns were being raised that GA was being overmedication at the placement; GA's family and IMCA were also raising other concerns about the placement.

The assessment completed in June 2021, "and the recommendation of the multidisciplinary team was that GA should move to a small residential home for adults with similar needs or to a supported living placement with support from staff with expertise in working with adults with learning disabilities, autism and complex needs" (paragraph 17). AA opposed this recommendation, and "maintained that her daughter should return to the family home and put forward an alternative care plan for the family to care for GA with some domiciliary support" (paragraph 18). However, "[t]he CCG confirmed that it did not propose to commission any domiciliary care or support in the event that GA was cared for by her family as they did not consider that the entire package of care met her assessed needs" (paragraph 19).

The court noted its position in considering these proceedings under s.21A Mental Capacity Act:

22. These are section 21A proceedings. The court's primary task is to determine whether or not GA meets the qualifying requirements in Schedule A1 to the Mental Capacity Act 2005: DP v London Borough of Hillingdon [2020] EWCOP 45. However, once an application is made pursuant to section 21A of the 2005 Act the court has broad powers to make declarations pursuant to section 15 as to whether P lacks capacity to make any decisions, and once such a declaration is made, the court has wide powers pursuant to sections 16 and 17 of the 2005 Act to make decisions on P's behalf concerning their personal welfare (CC v KK [2012] EWHC 2136 (COP); PH v A Local Authority [2011] EWHC (Fam)).

The court accepted "that an indication of GA's wishes and feelings is that when GA is with her family, she has on occasions indicated wanting to leave the placement with them at the end of their visits. Her relationship with her family and the mutual love they have for each other is a significant factor which would support GA returning home" (paragraph 31). The court further accepted at paragraph 33 that "GA's family are able to provide her with love and commitment, with familiarity, with a cultural and religious lifestyle in accordance with her background. I have no doubt at all that if she were returned to the care of her mother with the support of MA and other family members that they would love her and care for her to the very best of their ability." However, the court found at paragraph 31 that "there is no objective evidence to support the family's submission that separation from her family has had a significantly detrimental impact on GA's emotional well-being."

The court accepted at paragraph 32 that GA's family had raised 'legitimate concerns' "*in respect of some of the errors in the care provided to GA, the*

most serious being in respect of the Risperidone overdose and the failure to properly supervise GA and a male resident which resulted in an inappropriate encounter. I accept that such mistakes may happen in the proposed placement at *Placement B.*" However, on balance, the court found that the evidence showed the family had struggled to meet GA's needs "on a 24-hour a day basis, seven days a week. It is apparent from the documents before me that they have struggled at times to fully meet GA's needs. The family's plan relies heavily on the availability of the Day Care Centre and weekend support which the CCG is not prepared to commission in the event that GA returned home. I note that whilst being cared for by her family, the family were provided with a considerable amount of additional support through the day care and domiciliary care services, which would no longer be available. Even with this level of support concerns were increasing in respect of the family's ability to properly manage GA's needs and behaviour" (paragraph 34). The court did not consider that the family had addressed the concerns which had been raised regarding GA's care in the family home, and found at paragraph 36 there was "a clear discrepancy between the professional assessments in terms of GA's needs and how best to meet these, (examples being in respect of stimulation and engagement) and the family's understanding of those needs."

The judgment also included some of the evidence presented on the nature and intensity of GA's needs, both during the daytime and at night:

41. [...] GA requires a high level of support and supervision throughout the day to maintain her safety when engaging within day to day tasks. She has little danger awareness and within her current environment kitchen facilities are locked to ensure her safety. She will often throw items with little awareness of danger to self, others, or property. Throwing items does not appear to be in aggression but rather a play based, cause and effect seeking response for her. She requires support with her antiepileptic medication (potentially lifesaving) and with her complex sensory processing which in turn result in functional difficulties. She also needs support throughout the night to encourage a good sleep hygiene and monitor potential seizure activity. GA has Behavioural Dysphagia and this means that she is at risk of choking and aspiration by overfilling her mouth with food and drink, eating and drinking at a fast pace, vocalising with food and drink in her mouth and walking with food and drink in her mouth.

GA's MDT considered that GA's family would struggle to meet these needs, which

42. [...] would lead to an increase in the frequency and intensity of problem behaviours. This would in turn result in increased risks to both herself and to family members, particularly children. It is very likely that her medication would be increased if her environmental and relational needs were not met appropriately at the family home. A specialist home is able to provide a highly structured environment and teams of experienced staff who can provide the level of intensive interaction she requires whereas this would be exhausting and unattainable for a family unit.'

The court accepted at paragraph 43 "that the level of care needed is such care that would only be available from a professional team (not an individual). As Dr B indicates, during her time at Placement A, GA's dosage of Risperidone has reduced significantly, with her behaviour managed through careful structure and planning rather than medication, and the plan is that this medication can be ceased entirely. In all probability, in order to manage GA's behaviour in the home, the family are likely to require Risperidone to be reintroduced or increased. This would not in my judgment be in GA's best interests."

The court was satisfied that a move to Placement B was in GA's best interests, and any interference with her Article 8 rights was necessary and proportionate; plans were to be developed to *"enable*[...] *GA to maintain and enjoy a relationship with her family and the ability to celebrate important occasions"* (paragraph 46). The court did not adjourn the matter for consideration of trial at home or at Placement B, but relisted the matter to review *"following a period of settling in for GA"* (paragraph 48).

Comment

The case presents a useful snapshot of the type of case often argued in the Court of Protection but relatively rarely reported: one in which the statutory bodies consider, in the absence of any allegations of abuse, neglect or bad faith, that a person's needs are beyond the ability of their family to meet. The judgment is also notable as a slight counterpoint to *DP v Hillingdon*: while all parties agreed that GA should not continue to live at Placement A (which, strictly, would have been the subject of the s.21A challenge), the court looked to *CC v KK* as authority that it had 'wide powers' to consider her broader welfare.

The case is also of note in relation to the CCG's position that it would provide <u>no</u> care for GA in the event that she were to live with her family.

The judgment does not elaborate on this point fully, and it was noted that there had been some difficulty with the provision of care in the community, and at least one community care package had failed. Even in the absence of this important detail, we would observe two points in relation to this issue:

- In our experience, the position that a person must choose between a residential placement or no package of community care is one regularly taken by health authorities providing care under NHS Continuing Healthcare, and very rarely taken by local authorities providing care under social care legislation; and
- 2. Generally, when explored vigorously with them, CCGs will not maintain the position that their Continuing Healthcare duties are met by the mere offer of residential care if the Court of Protection indicates that it considers that the person's interests will be best served by living at their home or with family members, and typically will revise this position to make some offer of a community care package. The court cannot, however, **direct** the CCG to provide such a package, and the Supreme Court in N v ACCG made very clear that the Court of Protection should be careful before dedicating time to considering questions what might in the person's best interests in the abstract. The path for the parties - and the court - to tread can therefore be quite a narrow one.

Anorexia - why capacity matters

A Mental Trust v ER [2021] EWCOP 32 (Lieven J)

Best interests - medical treatment

Summary

In this case, Lieven J considered the capacity and best interests of ER, a 49-year-old woman with a diagnosis of anorexia.

ER's personal history with eating disorders had been a long and difficult one; the condition first emerged in her teenage years, and she had been treated in hospital from at least 2005. She had also had a very difficult personal life, and had served a custodial sentence for a number of years, during which time her anorexia worsened in severity.

From March 2012, ER had been repeatedly admitted to hospital as an informal patient, putting on some weight, and self-discharging quickly. The court noted at paragraph 6 that "[t]here is a pattern of her being unable to maintain anything close to a healthy weight in the community."

By 2021, ER had also developed serious physical health conditions, including 'renal failure as well as osteoporosis, endocarditis and klebsiella' (paragraph 7). Her treating doctor estimated that as a result of her renal failure (for which she had no prospect of receiving a transplant) her life expectancy was limited. She remained severely underweight following her last hospital admission in 2019, with a weight of approximately 35-37 kg. She had also recently overdosed on prescription medication.

The Mental Health Trust and NHS Trust sought declarations that ER lacked capacity to conduct proceedings and to make decisions concerning her anorexia; however, they were in agreement that ER <u>had</u> capacity to make decisions regarding her physical health. The health bodies also sought declarations that *"ER should not be*"

June 2021 Page 15

forced to accept treatment for her anorexia which she does not wish for, and that she should not be forced to go into a psychiatric hospital or a specialist eating disorder unit against her wishes" (paragraph 10).

ER spoke with Lieven J directly, who found that ER was "very articulate, clear in her views, and in my view, insightful as to her condition. I was concerned that given what I had heard directly, there was material that suggested that ER might well have capacity in respect of the two issues ligation decisions and decisions regarding treatment for her anorexia" (paragraph 11). As a result, the court heard oral evidence on ER's capacity in the relevant areas.

Capacity

The court readily accepted the conclusion of ER's treating clinician that she had capacity to make decisions in relation to her physical health conditions.

The issues in relation to her anorexia were more complex, and the court heard from both ER's treating psychiatrist (Dr P) and a psychiatrist offering a second opinion (Dr Cahill, who was also a specialist in eating disorders). ER's treating psychiatrist considered that

16...ER was unable to weigh up the information regarding the severity of her illness, and the impact it has had on her over the last years and in the immediate short term. She was also unable to understand the consequences of malnutrition and the risk that chronic low weight posed to her life. At that stage, Dr P considered that ER did have capacity to conduct proceedings.'

Dr Cahill's evidence noted that when he met with

her, "she was depressed [...] she said she had had enough, and felt that dialysis and medical treatment seemed a waste of time. He also records that she said she was very lonely, and that she thought that being around other people might make her feel better about life and improve her mood" (paragraph 19). He considered that ER was unable to weigh up decisions about her anorexia, and particularly, her inpatient treatment. The court recorded relevant portions of his evidence at paragraph 20:

she is not aware of her own disability. There is ample evidence that she lacks the insight into the seriousness of her condition when, at desperately low levels of BMI around 10, believing that a BMI of around 12.8 is safe. It is concerning that when at an incredible low weight, there is evidence of body image distortion, believing she is 'chunky.'

- While ER was able to understand and weigh up information regarding her renal disease, "there appears to be a lack of insight with regards to the physical health effects of poor nutrition and low weight. There is consistent evidence that [ER] believes she is eating enough to regain weight, despite evidence to the contrary."
- He noted that ER had never required nasogastric feeding or feeding under restraint – her inpatient treatment had functioned by helping her to be "in a much better place psychologically...we are not considering a 'traumatic' admission as such. If anything, [ER] is being looked after, with a reduction in isolation and loneliness."
- He considered that "at the heart of the condition is a fear of weight gain, a drive for

thinness, and a body image distortion. [ER] is likely to be fearful of this, whether she perceives this as relinquishing of control, a threat to her safety, security, and identity, or simply that she cannot tolerate the inevitable weight gain." ER stated that she did not "find the eating disorder units helpful, but gave reasons around it feeling military, regimented, controlled, and that the other patients were immature. Yet, through all the admissions, [ER] managed to restore weight, and was discharged at a more stable physical position than at admission. Although she talked about being watched in the bathroom as 'disgusting' which I do acknowledge, there was no evidence from [ER] that the admissions have been traumatic for her. She even recognised herself that she regained weight and came out 'stronger.' Therefore, it is very likely that her anorexic cognitions are driving her decisions regarding admission, and therefore, due to this impairment of the mind, in my opinion she struggles to weigh up the information."

The court also noted at paragraph 21 Dr Cahill's finding that ER "*has no desire to die and does wish to take steps to avoid that,*" which appeared to be in opposition to her rejection of treatment for her anorexia.

The court also considered the evidence of ER's solicitor, Ms Turner, who was very experienced in the field and considered that ER lacked litigation capacity. ER set out her own views:

22. [...] she does not like eating disorder units and sets out, in my view, perfectly rational reasons for this. In particular, that she is much older than other patients and finds their behaviours unhelpful for her condition. She also finds the approach of being encouraged to eat large meals very unhelpful to her.

ER expressed that she would like more support in the community, and felt that more support would be useful to her. She said she was interested in moving a residential placement or supported living, and that she felt lonely living on her own with visiting carers. She was clear she would not wish to be treated as a psychiatric inpatient, but did not want to die, and did want medical help.

The court noted the history of cases considering anorexia in the Court of Protection. Lieven J wrote that she considered the determination of capacity 'difficult' and even though there was consensus on ER's best interests, the issue of whether she had capacity was not academic and should not be 'fudged' with a view to the collateral impacts on ER:

31. [...] It is also right to acknowledge that it might strongly be in ER's interests to be thought not to have capacity as it allows the Court of Protection to have continued oversight of the case, which itself can provide more focus on the services that she needs. However, capacity and autonomy are such important principles, that lack of capacity cannot be assumed for the sake of expediency.

The court considered that ER was not a 'normal' anorexic patient. "Her renal failure is terminal, and she has a limited life expectancy, so the decisions she makes about not wanting an inpatient admission have to be seen in that context. Treatment would not prolong her life, therefore the views she expressed seemed potentially rational" (paragraph 32).

The court reluctantly accepted the evidence of the psychiatrists that ER lacked "capacity to make decisions about her anorexia treatment and, it follows, litigation capacity" (paragraph 33). Lieven J noted the views of Dr P, and observed that she would be "very slow to depart from the view of a treating consultant psychiatrist, absent any concerns about the closeness of the relationship, which I do not have here" (paragraph 33). The court was further fortified by the analysis of Dr Cahill, a clinician with "long and considerable experience of treating patients with anorexia nervosa and I wholly accept that is experience I do not have." The court accepted "that there is evidence of unrealistic thinking, especially around her weight levels" and "that there is evidence that ER does not act rationally in respect of some of the decisions she makes around her eating problems" (paragraph 33).

Best interests

In considering best interests, Dr P gave evidence that a best interests meeting had been convened, and it was agreed by all in attendance "that another inpatient admission against ER's wishes would not be in her best interests, given ER's strong opposition, and the fact that it is unlikely to have any impact on her renal position (so her physical condition is unlikely to improve significantly), but that it may impact on her mental condition significantly" (paragraph 17). The court agreed with the conclusion

34... [...] that it is not in ER's best interests for her to be forced to accept treatment for her anorexia which she does not wish to accept. In particular, she should not be forced to go into any inpatient hospital and treated against her wishes. In my view, it is plain that this is in her best interests given her renal failure and extreme dislike of eating disorder units and psychiatric hospitals. I also note that this conclusion accords with ER's wishes and feelings.

The court did emphasise that "*it is in ER's best interests to be given more support in the community' and 'the evidence is fairly clear that if she could be moved to a supported living placement where she can have dialysis and more support and company, this could much improve her mood and potentially improve her physical health over the next few months*" (paragraph 36). The court listed the matter for a review hearing and joined the local authority and CCG with directions to produce evidence on what support ER could be provided in the community, and consideration of a move to supported living.

Comment

The case is a useful reminder of that the obligation of the court to formally consider capacity and best interests are not vitiated because the outcome would appear to be the same whether the person were found to have capacity or not – here, ER would have chosen to forgo further treatment for anorexia, and her clinicians felt that she should not be obliged to accept it against her wishes. ER's capacity in this case was complicated by her serious physical health condition: a decision to forgo anorexia treatment during what appeared to be the last month or year or her life and which she found very unwelcome was one which to outward appearances to would also seem easier to understand. However, the court carefully considered both whether she had capacity, and where her best interest lay, coming up with a slightly different take to what had been proposed: namely, that further information was

required of the local authority and CCG to ensure she was being offered support on terms she was willing to accept.

Anorexia – the Court of Protection and the MHA

A Midlands NHS Trust v RD & Ors [2021] EWCOP 35 (Moor J)

Best interests – medical treatment

Summary

RD concerned a 37-year-old woman suffering from an extremely serious and debilitating condition, anorexia nervosa, since the age of 13. The NHS Trust sought declarations that:

- 1. RD lacked capacity to make decisions about her nutritional intake and about her care and treatment in general; and,
- 2. it was in RD's best interests to receive the care in her care plan, specifically that it was lawful not to take any steps towards forcing nutrition against her wishes, notwithstanding that, by so doing, it might in the short-term prevent her death.

RD had first been first admitted to an inpatient unit in 2000. Thereafter, there had been 14 admissions, approximately four of which were pursuant to the Mental Health Act 1983 on a compulsory basis. On most occasions, RD gained some weight but it was virtually all lost immediately following discharge. In 2019, there was a further significant deterioration in her physical health with possible vomiting (whether voluntary of involuntary).

During her admission in early 2020, the Trust commissioned a report from Dr Matthew Cahill,

a consultant psychiatrist. He agreed with the Trust that RD was suffering from a severe and enduring anorexia nervosa; her prognosis was very poor and her recovery was very unlikely. He recommended two further interventions: (1) discharge on a Community Treatment Order ("CTO") with four visits per day and (2) admission at a specialist rehabilitation unit. The second proved impossible because she did not meet the requirements of the unit – her BMI was too low; she did not demonstrate a willingness to engage and make changes; and she was not physically stable.

The CTO was put in place; but unfortunately, RD had to be readmitted to hospital in July 2020 and she was put on a NG tube. There was some physical restraint, but sedation and extended restraint were rejected due to the risks to her. There was no significant improvement in her BMI; and she was therefore discharged home. The application to court was then made.

It was agreed that she lacked capacity – she understands that, if she does not eat or drink her juices, it will not be good for her but she does not understand that it will lead to her death and that she needs to comply with the treatment.

The clinical evidence was that all viable inpatient treatment options had been exhausted. Any further detention under the MHA 1983 was likely to have a detrimental effect on her wellbeing and make the situation worse. The risk of forcefeeding under sedation or physical restraint far outweighed any likely benefit. There was a risk of choking on her own vomit or aspiration pneumonia. It would also cause extreme distress and psychological trauma and, in consequence, also significant physical harm as there was a risk of bone fracture and bruising. The treatment was causing more harm than good. Dr Cahilll agreed and considered that the treatment options were increasingly unethical and disproportionate if administered coercively. He considered that the case was 9 out of 10 on the seriousness scale.

RD agreed with the NHS Trust that, if she refuses treatment, it should not be forced on her, despite the possible grave consequences. She was clear that she wanted to live but that she did not want NG feeding and she did not want to go back into hospital under compulsion. She wanted to be at home.

Moor J observed that RD's life is undoubtedly a life worth living, but that the problem was she had been overwhelmed by her anorexia nervosa. He considered that the cycle of compulsory admissions had been distressing to her and that they had achieved very little. He accepted that everything should be done in the community to convince RD to take, voluntarily, the nutrition, treatments and drinks she needs. He decided to remove the threat of compulsion or compulsory admission from RD; and that he would remove the requirement to have NG feeds, unless she wished to have them. He urged RD to comply with the doctor's recommendations, but authorised and made the declarations sought.

Comment

The hope of all before the court was that RD would follow the doctor's recommendations,, but Moor J feared that if she did not, then she was likely to deteriorate quickly and would need palliative care.

The case also raised two important procedural points. The first in respect of the interaction between the MCA 2005 and the MHA 1983 in

cases such as this; and the second was whether the hearing should be heard in private.

The declarations were sought pursuant to both the MCA 2005 and the inherent jurisdiction of the High Court. Moor J cited Mostyn J in <u>Nottinghamshire Healthcare NHS Trust v RC</u> [2014] EWCOP 1317 in respect of the need for an application to court where a decision is taken not to impose treatment pursuant to s.63, where Mostyn J had held that:

In my judgment where the approved clinician makes a decision not to impose treatment under section 63, and where the consequences of that decision may prove to be life-threatening, then the NHS trust in question would be well advised, as it has here, to apply to the High Court for declaratory relief. The hearing will necessarily involve a 'full merits review' of the initial decision. It would be truly bizarre if such a full merits review were held where a positive decision was made under section 63, but not where there was a negative one, especially where one considers that the negative decision may have far more momentous consequences (i.e. death) than the positive one.

Moor J accepted that he should determine the application under both the MCA 2005 and the inherent jurisdiction notwithstanding the fact that he was undertaking a full merits review pursuant to the MCA 2005. That was because he considered that a full merits review under the MHA 1983 engaged public law matters, in particular, the safety of the public (although it is fair to note that it is not obvious that the safety of the public was central to the issues in RD's case). Alongside Moor J's main judgment, there was also a short decision [2021] EWCOP 36 in respect of the case being heard in private. He determined that in light of the parties' representations, particularly the express wishes of RD that the intensely personal matters should not take place in public, it was appropriate for the case to be heard in private.

Dialysis - how far is coercion justified?

University Hospital Birmingham NHS Foundation Trust v Al & K [2021] EWCOP 37 (Hayden J)

Best interests – medical treatment

Summary

In this case, the Trust sought a declaration from Hayden J that it was lawful and in Al's best interests to discontinue any further attempts to provide dialysis.

Al had a history of schizophrenia and was diagnosed on 30 September 2019 with end stage kidney disease. He had required long term haemodialysis to remain well but had only intermittently accepted treatment. He had a fixed delusional belief, as a consequence of his schizophrenia, that there was nothing wrong with his kidneys and he did not need dialysis. He also considered that the hospital was stealing his blood and he was accruing a large bill that he was unable to pay. This was causing him significant agitation.

It was not in dispute that AI lacked capacity to decide whether or not to have dialysis.

In July 2020, AI had required hospital admission, which included spending some time on ICU before being transferred to the Renal ward at Queen Elizabeth Hospital. He received haemodialysis twice a week, verbally refusing each time but when the nurse arrived at his bedside, he would passively let them connect him to the machine using his dialysis line. No sedation or restraint was required.

The stage was reached that AI was physically well enough to be discharged but continued to state that he would not attend his outpatient dialysis appointments. The clinical view was that it was in his best interests to be discharged back to his care home and should not be compelled to attend and receive dialysis. Instead, the Trust would act "reactively", which meant that a vehicle would be sent to AI's home on the day that dialysis was required and it was left to him as to whether he would comply or not. His family took the view that AI should be compelled to attend and receive dialysis. Hayden J agreed with the clinicians that chemical and/or physical restraint was not in AI's welfare interests.

An application was made to the Court of Protection. Theis J made the first order sought on 23 November 2020, including that AI lacked capacity to make decisions about his treatment in relation to dialysis for End Stage Kidney Disease. She declared that it was lawful and in AI's best interests to be discharged from Queen Elizabeth Hospital, to receive reactive treatment for dialysis for End Stage Kidney Disease, and not to be compelled to receive dialysis by any form of restraint.

After his discharge on 11 December 2020, he voluntarily attended a number of sessions of dialysis, but then he did not attend again from 12 January 2021. His health seriously deteriorated and he required further hospital admission. He cooperated with dialysis in hospital but was again then discharged. This pattern continued.

His non-compliance Haled to fluid volume overload, which resulted in severe breathlessness. His physical deterioration had also significantly worsened.

Al was admitted to Accident and Emergency on 18 May 2021 with breathlessness. He accepted oxygen. He initially declined dialysis, following which he agreed to it for a short period of time before pulling his tunneled line out. Eventually, he was persuaded to accept the insertion of another temporary line.

Dr Hewins, the consultant neurologist, was extremely concerned by his presentation. There was prolonged bleeding from the groin where the dialysis was inserted. He was at risk of heart failure and significant risk of death over the next 24-48 hours. Even if his blood pressure were to stabilise, the clinicians could attempt to insert another tunnel but the likelihood of being able to achieve that was significantly uncertain. It would be a potentially painful and distressing procedure. Dr Hewins' opinion was that the clinicians were in danger of pursuing inappropriate offers to re-establish dialysis without any realistic likelihood of durable benefit when the focus should be on palliative care.

Hayden J observed that AI had been consistently resistant to cooperating with dialysis. He had only passively cooperated when he was physically weak and struggling to breathe. He then disengaged when he felt better. He noted that AI's superficial cooperation with dialysis had gradually reduced and that on his most recent admission, he had withdrawn the tunnel line, which would have been painful. Hayden J drew the inference that AI has become tired by the effort of dialysis. He had also become neglectful and disinterested in his appearance, which was contrary to his personality.

The neurologists considered that if AI was administered further dialysis he would last for a maximum of 2 to 3 weeks; and that the process of dialysis might precipitate collapse and death.

Hayden J also heard evidence from Al's family to ensure that the court was considering his welfare in the widest sense. They described his upbringing in Pakistan and that he was sociable, as well as his enjoyment of spiritual Pakistani song and food. He had recently become a grandfather and was delighted by that. He also attended all family events.

As to prognosis, Hayden accepted that AI was at the end of life; and therefore, the question was determining how, over what period and in what circumstances, he died. He endorsed the palliative care plan because he considered that it conveyed dignity on AI at the end of life and granted the declarations sought.

Comment

This case has interesting resonances with that of *RD* also covered in this issue. In both, it would be possible to construct coherent legal arguments that coercion **could** occur, but the real issue was whether it **should**.

It is worth remembering, in the context of the best interest assessment, that section 4(6) MCA 2005 requires the decision-maker to consider the person's past wishes and feelings *"so far as is reasonably ascertainable"*. In this case, that exercise involved Hayden J forensically analysing Al's behaviours in response to this dialysis in order to understand his wishes and feelings. Hayden J observed that *"behaviour, when assessed carefully, may sometimes* *communicate feelings more effectively and accurately than words."* Ultimately, as Hayden J acknowledged, the court is charged with the protection of AI's autonomy, and his wishes and feelings are integral to that. He cited in this regard his own judgment in *SS v London Borough of Richmond Upon Thames & Anor* [2021] EWCOP 31, covered further also in this issue.

Securing autonomy – Article 8 ECHR in the balance

A Local Authority v TA & Ors [2021] EWCOP 22 (Cohen J)

Mental capacity – best interests – contact – residence

Summary

The contours of this difficult case were sketched out economically in the opening paragraphs of the judgment:

1. This case concerns the health, welfare, and care of GA, an 87-year-old lady. She is a widow with six surviving children three girls and three boys - and there are 10 grandchildren, by two of the girls. Since 2004, she has lived at an address in a city in West Yorkshire. She is the tenant of that property which is a housing association property. Of her six children, at all material times, two of the boys, TA and HA, have lived with her.

2. HA has some form of mental disability and is in receipt of a care package, the details of which are not known to me or, I think, to the social workers dealing with GA, but I have no reason to think that either (a) he presents any risk to GA or (b) he is able in any way to assist GA in her care. 3. TA is plainly an intelligent man and he presents himself as his mother's sole carer and the proprietor and manager of a care home - an inappropriate description of his mother's home but one which he says should entitle him to £1 million a week by way of salary, for the care that he says that he provides to GA and HA.

4. It is the case of the local authority that TA exercises abusive and controlling behaviour towards his mother and so dominates her life that she (i) is unable to enjoy personal dignity; (ii) has lost contact with her community and with her family, apart from HA and TA, and to some extent XA [one of GA's daughters]; and (iii) is denied access to important healthcare and treatment.

The local authority sought to remove TA from GA's home and bar him from returning so that a local authority care package could be put in to look after her.

TA had previously been the property and affairs attorney for his mother, but had been removed in 2018, having been convicted in February 2016 of fraud five counts of and abuse of position. Those five counts related to money that he had misappropriated, that money being due to or for the benefit of his mother, and she was the subject of three of the five counts. TA had been sentenced to a term of 45 months' imprisonment. TA asserted that his conviction was wrong and that he should never have been charged but had not appealed, and blamed the local authority for wrongful prosecution of him.

There was an extensive procedural backstory to TA's non-attendance before the court,

notwithstanding the fact that he had been ordered to attend. The backstory included publication of meetings involving professionals on YouTube and this judgment from Cobb J addressing TA's wish to record proceedings (and also his conduct towards court staff).

There being no doubt as to GA's lack of capacity to make the material decisions, Cohen J was squarely confronted with determining what was in her best interests. Despite his nonattendance, TA's case was squarely before the court, and the most economical way in which to outline how Cohen J reached his decision is to set out how he responded to that case:

71. I agree with the local authority that what is being sought is draconian and I have considered it very carefully. I should return to the concerns of the local authority and the Official Solicitor. I deal with them under these headings. First, they say that GA is at risk of immediate harm. TA goes out, as he agrees, leaving GA, and it is presumed HA, shut in the house. This happens most days for about an hour and they are there on their own. If there was a fire in the house, or GA suffered a stroke, the result could be fatal. TA says that he can tell what is going on in the home by looking at his mobile phone and by picking up the remote surveillance but that, of course, is no substitute for presence and is inherently dangerous.

72. Secondly, the local authority argue that GA is deprived of many of her basic rights by TA. She is isolated; she does not go out; and she remains in her own room. Neighbours report to the local authority that they have not seen GA since 2019 and that previously, she was a sociable woman. It is not known whether she is, in fact, able to go out and enjoy being outside and seeing other people. That simply does not happen and cannot be investigated at the present time.

73. Thirdly, she is deprived of medical attention. As I have already mentioned, there have been occasions when nursing teams have been prevented from entering the house and there is no record of any GP attendance either at the home or at a surgery, so far as the local authority has been able to find, since 2019. At aged 87 and having had a heart attack and been the subject of a whole series of heart medications, it is plain that she should be receiving medical attention.

74. Fourthly, there is no evidence that GA is receiving any form of medication. TA refused to provide any information. I do not know why. Maybe he feels information is power. However, the result is that, so far as the local authority and the court are concerned, there is no evidence that GA is receiving statins, blood thinners, and her heart medication and there is no evidence that there have been any of the necessary tests undertaken to ascertain if the medication previously prescribed is still appropriate.

75. Fifthly, she is cut off from at least three of her children and all of her grandchildren. Whatever TA may say about their failings, RA, MA, and DA all feel prevented from seeing their mother. MA and DA are the only ones with children. So GA does not see her grandchildren at all. GA wants to see them but they will not come and see her with TA running her home.

76. Sixthly, all her intimate care is carried out by TA, when she would much prefer a

June 2021 Page 24

female to do that. It can only be degrading for her to be washed, bathed and have her pads changed and cleaned after using the toilet by her adult son. For a woman of her faith [i.e. Muslim], it must be particularly humiliating.

77. Seventhly, she is under video surveillance 24-hours a day. I accept, of course, that in some medical settings that happens, but it is degrading to anyone's sense of privacy and cannot be justified for a lady who seems to spend nearly all of her hours in bed.

78. Finally, she is deprived of any form of professional assistance. Those who offer essential services simply will not go into the property because of TA's dominating and controlling presence and his insistence on filming what goes on.

In the circumstances, Cohen J identified that in making the best interests decision on behalf of GA, it was common ground that:

80. [...] above all GA would want to stay in the home in which she has lived for some time. I am sure she would want to receive appropriate medical treatment and medication. She would want to be able to see her doctor. She would want for example, to receive treatment for the pain she demonstrated she was suffering on 26 and 27 January. She would want to see her children, all of them, and her grandchildren. That is what she said. She would want to be cared for by women. She would want care and stimulation. All these are very much in her interests. I am sure also that she would want to be cared for by those who know her and from whom she is used to receiving care.

Cohen J had to consider TA's rights: TA lived in

the property and no other home of which he was aware. Cohen J did not know where he would live if required to live – although observed that it was TA's fault that he was without that information. He also observed that to deprive him of the company of his mother was a significant interference in his family life. In the circumstances, and had there been any way of enabling care to be shared to make good the deficits of the care GA was receiving from TA, Cohen J made clear (at paragraph 82) that he would have grasped it. The problem that he was satisfied that:

82. [...] the history, from the evidence, and from the argument, that that is simply not possible. The video clips make it very clear why it is not possible. There is no suggestion from TA, even in the video clip that he sent today, that anything will change. He does not see a single deficit in the care that he provides and I come to the clear view that a different regime must be put into place in GA's best interests.

Cohen J expressed the hope that, as time could go on, TA could be brought back into the fold and able to see his mother and spend time for her, as he was sure that would be to their mutual benefit, but:

83. [...] a continuation of the current situation, where his mother is locked away by him from the world and the rest of her family without others being able to reach her, except occasionally with the most stringent court orders, is not a situation that is in her best interests. I accept that this is a draconian order. It should have been capable of being avoided but it is TA, and TA alone, who has brought this about. I therefore accept the local authority plan must be one that would be implemented.

Cohen J found that the interference with TA's rights were necessary and proportionate because there was no other way of promoting his mother's rights and interests (paragraph 85). He also accepted that the provisions within the care plan amounted to a (justified) deprivation of GA's liberty.

The order sought by the local authority - and endorsed by Cohen J - provided for the immediate vacation of TA from the home (the tenancy being in his mother's name, and TA being only a licensee there, Cohen J did not have to grapple with the complexities which would have arisen if he had his own rights in the property). The local authority had undertaken to provide TA with fully funded accommodation in a bed and breakfast for 14 days. As Cohen J had not heard from TA at all as to what his proposals might be and whether or not he can go and stay with other members of the family, he considered he could not nothing other than endorse the proposal for 14 days. The order also sought the removal of XA from GA's home; Cohen J considered that XA would no doubt return to home in Greater Manchester.

The order also contained a cordon put around the property of about 100 yards by reference to six named streets in which TA was forbidden from entering. The local authority did not want him watching the house or monitoring who goes in and out. He was barred from removing the Motability car from the property which is there for his mother's benefit. There was a further prohibition about him putting articles or other information in the public arena, including on a social platform, and Cohen J considered that was plainly needed in light of the history of the case. Cohen J also made an order limiting TA's ability to correspond with the local authority and with the Official Solicitor.

Cohen J made a civil restraint order against TA for a period of two years, noting that: "there have been four occasions before this hearing began when applications have been dismissed as totally without merit, all of them within the last vear. In addition, I have dismissed three applications as being totally without merit in the course of this hearing. There are another four recent occasions when applications have been dismissed as showing no reasonable grounds or no good reason. The threshold for the making of a civil restraint order is plainly crossed. I therefore will make a civil restraint order as the only way to restrict the level of applications. The local authority no longer pursues an extended civil restraint order and there will be a civil restraint order for a period of two years" (paragraph 91). Cohen J made provisions for a further hearing further to consider questions relating to deprivation of liberty, contact, and also committal proceedings which were on foot in relation to TA

Comment

Whilst not establishing any new propositions of legal principle, the case serves as a reminder of the stark issues that arise when the Court of Protection is grappling with what (in effect) are adult care cases. As noted above, it made Cohen J's life considerably easier that he did not have to consider how to navigate a position where TA had his own rights in the property, a feature of many such cases. Given the history of the case, it is overwhelmingly likely that Cohen J would have found some way in which to remove TA, but it would have made his (necessary) balancing of XA and TA's rights even more challenging.

COVID-19 vaccination – best interests is not code for mandatory vaccination (and a reminder you can't fail a capacity assessment)

SS v LB Richmond on Thames and SWL CCG [2021] EWCOP 31 (Hayden J)

Best interests – medical treatment

Summary

This vaccination decision is different to others that have been reported to date (such as E (Vaccine) [2020] EWCOP 14, because it was clear that the objection to the vaccine came from the person themselves, not from family members. SS, the person in question, was 86, and had a diagnosis of dementia. Proceedings had been brought on her behalf under s.21A challenging her placement in a care home. During the course of those proceedings, it became clear that SS was increasingly resistant to medical interventions of any kind, although, as Hayden J noted, it was not clear whether this was due to her advancing dementia or was a longer-term feature of her personality and general approach to life is unclear.

The care home in which SS lived was a large one, 28% of whom (27 residents) had died during the first lockdown. Since January 2021, a GP, Dr N, had been visiting on a regular basis to administer initial and follow-up vaccinations to residents and staff. She had regularly spoken with SS to try to encourage her to have her vaccination. This has been reinforced by the care staff. Dr N conducted a capacity assessment with SS on 23rd March 2021, she noted:

"patient appears confused and unable to fully understand. Initially agrees to have vaccination and reports will go down the road to her usual doctor and get it. Subsequently refuses the vaccination. Carer.. re explained what I had relayed to the patient and the patient has difficulty understanding the reason for the call and the vaccinations. Doesn't appear to understand what Coronavirus is despite being explained to twice. Unable to retain any information given. Patient then declines continue with to the consultation. Patient failed capacity assessment as unable to fully understand the information given nor retain the information [SS] doesn't have capacity to consent for Covid-19 vaccination.

In the event of a best interest meeting, I would support vaccination and it would not be necessary to review the patient's capacity at the time of administration of vaccine given that her dementia is a progressive condition since at least 2019 when it was last formally assessed."

At paragraph 13, Hayden J observed that The ambit of this assessment strikes me entirely consistent with as that contemplated in: SD v Royal Borough of Kensington And Chelsea [2021] EWCOP 14; E (Vaccine) [2021] EWCOP 7. I hope Dr N will not think me too pedantic if I make the observation that "patient failed capacity assessment" strikes me as awkwardly expressed. It is not a test that an individual passes or fails, it is an evaluation of whether the presumption of capacity has been rebutted and if so, for what reason.

An application was made to court following discussions between SS's Accredited Legal Representative and the local authority, and Dr N provided a more detailed assessment, in which she noted that the:

administration would be challenging with this patient. Any physical restraint would need to be necessary and proportionate with the minimum amount of force for the shortest period of time. I believe restrictive physical intervention will be required in this case to protect the patient and staff from coming to harm. Clinical holding by several staff members should be sufficient for administration in this case. However, this would be a decision taken jointly with the vaccine administrator and care home staff."

Under the sub-heading 'belief structure,' Hayden J set out his understanding of the picture of SS's beliefs about medical treatment:

18. SS is recorded as having been compliant with her medical regime when she first arrived at the care home. However, as has become clear from several sources, there came a point when she discovered a newspaper article which she read as arguing that medicine "did more harm than good". Quite when she first came by this article is unclear. In any event she had not been at the care home for very long before she started to brandish it, with characteristic forcefulness, at anybody who sought to afford her medical care of any kind. The consequence has been that SS has been almost entirely non-compliant with any attempted intervention. Ms Kelly Fisher [team leader at the care home] told me that some attempt had been made to administer covert medication to keep

SS's blood pressure more stable. However, she said SS was "like a blood hound" who could tell something was amiss and this plan was abandoned. I hasten to add that she made this remark in an affectionate way. Indeed, I was struck by the extent to which all Ms Fisher's remarks about SS were both kind and respectful.

19. In June 2020 SS developed a rash on her legs. Even though this only required the application of cream SS resisted steadfastly. Staff and medical professions made great effort with her to encourage her to receive the treatment, but the outcome can only best be described as having "limited success".

20. In her report, (see para 13 above) Dr N reviewed the medical records and noted that there is no record of SS receiving any vaccination of any kind at all. The records go back to 1997. Even more strikingly there is an unambiguous note that SS declined both seasonal influenza and pneumococcal vaccines when offered them by the surgery. The first of these refusals is recorded in 2002 with entries identified in 2010, 2012 and 2014. Thus, there is a clear and consistent pattern of behaviour which predates SS's diagnosis of dementia by a significant period.

21. As Ms Hancock, counsel for SS, points out, whilst the above behaviour gives a strong indication of what SS's capacitous wishes and feelings might be on the question of the Covid-19 vaccination, that must still be placed in the context of medical records which signal a history of co-operation and engagement with medical professionals. The notes prior to 2015 reveal SS to be a woman who is responsible and proactive in her treatment. Thus, there is a pattern of routine blood tests every few years; vital signs checked and monitored annually by the GP surgery; she has attended walk-in clinics, out-patients hospital clinics (on six occasions, with various complaints, between 2002 and 2010). It is also notable that she underwent a series of sometimes intrusive investigations between 2004 and 2009 e.g. endoscopy, x-rays and ultrasounds. As the dementia took hold, SS was noted to have presented to her pharmacist seeking to collect her prescribed medications before finishing the previous prescription. It seems likely that this behaviour reflected her deteriorating short-term memory. Reinforcing this is a changing pattern, SS after 2015, when misses appointments, fails to pick иp prescriptions etc.

Hayden J then set out in some detail the analysis of the risk faced by SS, which merits reproduction in full as largely representative of the risk faced by many residents in care homes as at April-May 2021:

22. Apart from her cousin, TB, who has, in the past, visited approximately 3 or 4 times per year and during the period of social restriction spoken to her occasionally by telephone, SS receives no visitors at all. She is reserved and private in her approach to life and temperamentally inclined to keep her distance from others. She is at very low risk of infection from the other residents, all but one of whom has been vaccinated.

23. As care homes finally open up to more visits from family and friends, an identifiable risk is presented which has to be negotiated. This large care home makes provision for compulsory lateral flow test to visitors, many of whom will themselves be fully vaccinated. There is a further risk presented by staff members. The team leader in the care home has told me that 77 of the 100 members of staff have been vaccinated. Of the 23 who have not been, a few have declined for recognised medical reasons. The remaining individuals resist the vaccine in principle, some believe that it is, as yet, insufficiently tried and tested. By this, as I understand it, they are contemplating some unidentified adverse reactions which have yet to be exhibited.

24. Certainly, nobody could sensibly doubt the efficacy of the vaccination programme. The National Health England statistics, almost daily updated in the public domain, tell their own explicit success story. A few weeks ago, Covid-19 was reclassified as 'endemic' in the UK i.e. no longer 'pandemic'. The forthcoming months generate cause for optimism but without any guarantees. The need for booster vaccinations and/or vaccinations modified to combat Covid variants remains a likelihood. Thus, the greatest risk, statistically, to SS comes from unvaccinated members of staff circulating in the community away from the care home. Properly recognising this risk Ms Christine Fisher [the care home manager] tells me that all staff members are required to have two lateral flow tests per week and one Polymerase Chain Reaction (PCR) test. The extent of the weekly testing reflects the calculation of risk. Having manifestly given the matter very deep consideration, Ms Fisher told me that ultimately, though she would wish all staff to be vaccinated, she considered the decision to have the vaccination to be an exercise of personal choice. (emphasis added)

Page 29

June 2021

A further indication of risk was provided by Dr Prabhakaran, a psychiatrist instructed to give expert evidence as to SS's capacity. He summarised the position as follows:

8. General infection risk in care homes in England has reduced considerably due to various factors, as below.

- Covid infection rates are currently very low in England.
- There has been a high level of vaccination uptake amongst care home residents and health and care staff. Vaccines have been effective with studies demonstrating that healthcare staff are 86% less likely to develop infection after two doses of vaccine.
- Sero-positivity rate in age group 70-84 is high at 98.4 in weeks 12-15 2021(April data – see attachment)
- Effective use of PPE. And other infection prevention and control measures.
- Effective test and trace including use of rapid lateral flow testing visitors to care homes.

9. However, SS would be considered 'clinically vulnerable' based on Public Health England criteria primarily due to age, presence of advanced dementia and other medical conditions such as hypertension and high cholesterol. Although current infection levels remain low, residing in a care home environment confers a higher risk of being infected with coronavirus compared to being in a private residence. There are concerns of a third wave of Covid 19, which would place SS at higher risk. In the circumstances, it was clear, Hayden J found that SS lacked capacity to decide as to vaccination, such that:

25. [...]The issue for the court therefore is one of determining what is in SS's best interests. Having regard to the available evidence I consider that there is substantial material from which to conclude that SS if capacitous would most likely have declined the vaccination. Though she attended quite fastidiously to her general medical welfare, she plainly resisted vaccinations. The evaluation of what SS would have wanted is in this case, inevitably imperfect. Capacitous individuals facing a frightening pandemic might very well take a different view of a vaccination which restores them to their liberty than, for example, a decision not to take a flu vaccine. Ultimately, the forensic tapestry can only be woven from the available thread. However, it must be borne in mind that even though a capacity to weigh and balance the decision in focus has long disappeared, SS has nonetheless consistently and volubly opposed the vaccination. SS's reality is undoubtedly delusional, but that does not stop it being her reality. This has to be both recognised and respected.

Hayden J considered, and rejected, a suggestion by SS's cousin that she be told that her father, who was long dead, but who everyone understood to be very much alive in her mind, had requested that she take the vaccination. At paragraph 32, Hayden J identified that:

32. [...] This involves feeding into a delusional belief system. Whilst that may occasionally have been necessary in negotiating routine day to day challenges, it risks, in this context, compromising all

involved. It requires there to be a collusion to trick SS into complying with a vaccination which, on balance, it seems unlikely she would have wanted whilst capacitous and certainly does not want at this point. It is an artifice of a different magnitude and complexion to those earlier more mundane negotiations. It becomes disrespectful to her, not merely as the woman she once was but to the one she is now. Though undoubtedly a well-intentioned suggestion, it risks compromising her dignity and suborning her autonomy. It cannot, in my judgement, be in her best interests. I entirely understand TB's instinctive view that such means might justify the end, given the protection that the vaccine would afford SS. I hope he does not read my reasoning above as, in any way, intended to be a criticism of him. It most certainly is not.

In the face of a submission by the CCG that it was in SS's best interests to have the vaccination, Hayden J noted that he would have no doubt that this would be correct:

36. [...] were I to confine the issue solely to the health-related states, events and data he identifies. A determination of "best interests" in this context however is, for all the reasons discussed above, not to be confined to the epidemiological; it requires evaluating welfare in the broader sense. As Baroness Hale said [in Aintree], it requires us to put ourselves in the place of the individual concerned.

In SS's case, it was clear that there:

37. [...] was no question of SS being supine or passive if she recognised that the vaccination was being given against her will. One of the carers noted that those involved in attempting any "gentle restraint" had better be "kung fu experts", as she put it. The plan which involves both sedation and restraint contemplated the carers' involvement. Ms Fisher did not think that was appropriate. She told me that she thought that SS would look to her carers for help. They would not be able to intervene; that would be distressing for both parties. Moreover, in Ms Fisher's analysis it would most likely dismantle the tentative trust that had been established over the months and in consequence of sensitive and determined professional effort. I find this reasoning to be measured and persuasive. The Local Authority and the Accredited Legal Representative on SS's behalf both submitted that when evaluating welfare in the broader sense, it could not be said to be in SS's best interests. Lagree.

On a procedural note, Hayden J noted that cases relating to COVID-19 vaccination are now usually heard by Tier 1 or Tier 2 J judges (for an example of the latter, see <u>Re CR</u>), but that this had been allocated to a Tier 3 judge because it presented "for the first time, an opportunity to evaluate strongly and consistently expressed views by P relating to vaccination and the weight they should be given, in the broader landscape of the insidious risk arising from the Covid-19 public health crisis" (paragraph 35).

Comment

This decision serves as a strong reminder that decision-making in relation to COVID-19 vaccination must be undertaken on an individual basis where the person lacks capacity, and that the answer is not always that it is in their best interests. It also serves, more generally, as a

summary both of the (devastating) ravages of the pandemic in care homes, of the risk matrix as at spring 2021, and also a reminder of where risks are most likely to come from in the case of those in care homes who do not have visitors.

More broadly still, Hayden J's observations about best interests decision-making in relation to a person with beliefs that can properly be described as delusional are important - that a person may be living in a reality entirely different to that experienced by anyone else does not mean that is not their reality, around which the decision has to be constructed. They chime also with the earlier observations of Peter Jackson J in the Wye Valley case that it was "no more meaningful to think of Mr B without his illnesses and idiosyncratic beliefs than it is to speak of an unmusical Mozart." And Hayden J's observations about the dangers - for all - of embarking upon a path of deception are powerful (even if there may be cases where such could be justified: see, for example, Re AB [2016] EWCOP 66); for more on the importance of truth-telling in relation to dementia, see the important 2016 Mental Health Foundation report: What is truth? An inquiry about truth and lying in dementia care | Mental Health Foundation.

Lastly, Hayden J's observation about the (familiar) use of the phrase that SS "failed her capacity assessment" is an important reminder that there's no such thing as failing an assessment. An assessment is the process of thinking, with an open mind, about the person's decision-making capacity (and providing them with all practicable support along the way); that assessment continues for so long as is required, and including gathering such external evidence as necessary - e.g. observation of the person in

action - until it is possible to reach a conclusion as to whether or not the person has capacity to make the decision in question.

Sex and the limits of fluctuating capacity

Re DY [2021] EWCOP 28 (Knowles J)

Mental capacity – sexual relations

Summary

When and how does the concept of fluctuating capacity apply in the context of sexual relations? This was the question confronting Knowles J in this case. DY had just turned 18; she had been diagnosed with two chromosomal duplicities: fetal alcohol spectrum disorder, and a moderate learning disability. She had also been diagnosed with developmental trauma disorder or complex post-traumatic stress disorder. The applicant local authority had had responsibility for her as a looked after child since 2012 and had ongoing responsibility for her adult care services. The local authority initially sought authority to deprive DY of her liberty under the inherent jurisdiction of the High Court (DY at that point still being under 18). The case ultimately came before Knowles J, at which point it was common ground that DY lacked the capacity to conduct the proceedings and to make decisions about care, contact, social media use and her finances. However it was accepted that she had capacity to decide between the two residence options which were capable of meeting her assessed needs. DY was, at the time of the hearing in a "boyfriend/girlfriend" relationship with a man, AB (who did not have learning difficulties).

The issue in dispute was DY's capacity to decide to engage in sexual relations (Knowles J noting that the Court of Appeal had 're-cast' the test in <u>JB</u>). The positions adopted were:

- On behalf of the local authority, DY was able to make a capacitous decision when settled or in a familiar situation or surroundings, but might be unable to do so when unsettled or distressed;
- On behalf of DY by her litigation friend, the Official Solicitor, that she had capacity, and that the independent expert, Dr Claudia Camden Smith, might have set the bar too high in her analysis of DY's understanding of the distinction between consenting to sexual relations within and outside a relationship.

Knowles J summarised Dr Camden Smith's final position (after giving oral evidence) thus at paragraph 20:

In her oral evidence, Dr Camden Smith conceded that she may have set the bar too high in terms of her analysis of DY's understanding of the distinction between consenting to sexual relations within and outside a relationship. Whilst she accepted that DY had capacity to consent to sex. her concern was how DY would make that decision outside of a relationship. It was difficult to say what her capacity would be if her relationship with her present boyfriend were to end. When unsettled, DY may be unable to make a clear and rational decision. DY did not think the sex act was very special and had a pragmatic unemotional view of sex that could be something quite transactional rather than something particularly intimate because of her experiences growing up. Dr Camden Smith accepted that her view that DY thought sex was not special had not been specifically explored with DY but was

based upon DY's presentation. She confirmed that DY understood the mechanics of the sexual act, the risk of pregnancy, and the risk of a sexually transmitted disease. DY had told her that she had said no to sex within the context of her relationship with her boyfriend and that she had had penetrative vaginal intercourse with her boyfriend. The main confusion arose because DY could not conceive of not being in a relationship with her present boyfriend and the nature of DY's cognitive deficit meant that it was much harder for her to analyse things in abstract terms. It would be possible to support her if she expressed a wish in future to have a relationship with someone else. There were times when, unsupported, DY would lose capacity but if she were provided with support then her capacity would not fluctuate.

At paragraph 22, Knowles J acknowledged the "real tension" between:

a desire to protect DY and a decision to permit her freedom to engage in sexual relationships which might place her at some risk. Whilst the MCA 2005 and the case law warns me against losing sight of the fundamental principle that the obligation to protect the incapacitous must be tempered by respect for the autonomy of those with mental disabilities/disorders, my personal experience of meeting DY served to highlight her vulnerability. She was eager to please and found it hard to identify any risks she might face either generally or in relation to the issue of consent to sexual relations. When coupled with an awareness of her difficult personal history, I found myself concerned about her vulnerability in general and acutely conscious of how easy it would be to exploit and harm her. My experience of DY is, I believe, shared by those who have daily contact with her and those who are responsible for her care. Their anxieties about DY shaped the local authority's position in these proceedings and, whilst it was unfortunate that this was not initially as clearly articulated as it might have been, I do not criticise the local authority for taking the stance it did.

The parties were agreed that DY had capacity to engage in sexual relations when she was not upset or distressed, the issue in fact being the narrow - but important - one of whether she had that capacity when she was unsettled or distressed. The local authority - having been required by Knowles J to specify exactly what they sought - sought a prospective declaration to that effect or, alternatively, a declaration in identical or similar terms pursuant to the inherent jurisdiction. The Official Solicitor submitted that the local authority's approach was wrong in principle and wrong on the facts because (a) the court was required to assess capacity on a general and non-specific basis: (b) the evidence before the court could not rebut the statutory assumption that DY had capacity on that basis; and (c) any concerns about her vulnerability or ability to assess risk could and should be addressed through provision of support and best interest decisions on care and contact.

Rejecting the approach urged upon by the local authority, Knowles J identified that:

25. The law requires that capacity to decide to engage in sexual relations should be assessed on a general nonspecific basis. The local authority asserted that its formulation did not fall

foul of that requirement because it related to "circumstances as opposed to a particular person". I note however that the requirement to assess capacity on a general basis and the policy underpinning applies to the timina and it circumstances of the decision and not just to the identity of the sexual partner. The passage in <u>IM v LM</u> quoted above [16] makes plain that assessment of capacity to consent to sexual relations can only be on a general basis rather than tied to the specific prospect of a sexual relationship with a particular individual in specific circumstances. Likewise, Hedley J in <u>A Local Authority v H [2012] EWHC</u> 49 (COP) held that capacity had to be decided in isolation from any specific circumstances of sexual activity.

26. Absent from the local authority's written submissions were any standards by which DY's level of distress or unsettledness should be judged to determine whether or not she had capacity to engage in sexual relations. Though Mr Foster suggested in his oral submissions that plain words such as "upset",

"unsettled" and "distressed" would be sufficiently descriptive and that it might be open to the court to provide greater specificity in that regard, I found myself unpersuaded by that submission. Whilst all those particular words might describe DY's mood at any given time, they did not adequately describe the point at which she might cease to have capacity to engage in sexual relations. As I understand the local authority's case, that would seem to be a matter of degree. difficulty with Mr Foster's The submission was that, on each occasion that DY appeared to be unsettled or distressed and was proposing or had the

June 2021

Page 33

potential to engage in sexual relations, the prospective declaration sought would require an assessment of whether DY was sufficiently distressed or unsettled so as to have lost capacity to engage in sexual relations. That would give wide discretion to individual professionals without any check to ensure that DY's autonomy was respected, and that decisions were not being driven by the desire to protect her.

27. Whilst I acknowledge that prospective declarations of incapacity are permissible pursuant to s 15 MCA 2005, these are exceptions to the general approach. They may be appropriate in cases where there is clear evidence of the circumstances in which a person would or may lack capacity in the future and where there were practical reasons why a declaration or declarations should be made in advance. Neither of those conditions applied in this case. Here, the distress and unsettledness were not well defined and, even if DY did experience such emotions, it could not be assumed that this would impair her decisionmaking ability without an analysis of the particular facts pertaining at the time.

Knowles J was clear that she had heard no evidence that would justify an order in the terms sought:

28. Dr Camden Smith's written report did not assert that DY's capacity fluctuated. Although she conceded that, if DY's relationship with AB ended, it might be difficult to say what her capacity was in those circumstances and that, when unsettled, DY may be unable to make a clear and rational decision, Dr Camden Smith did not say that DY would lack capacity applying the relevant statutory criteria. Even if DY's capacity were to fluctuate as was suggested by Mr Foster in guestioning, Dr Camden Smith was clear that, if DY were provided with support, she did not think her capacity on this issue would fluctuate. DY's social worker agreed with me that it was very difficult to tease apart how DY might respond to someone other than AB and conceded that she was speculating about what might happen in other situations. At its highest, the local authority's concern that DY may lack capacity to make decisions about engaging in sexual relations when distressed or unsettled was based on speculation as to how DY would respond in circumstances which may not even arise if care and contact were appropriately managed.

The local authority identified a wide range of generic factors about DY's presentation and vulnerabilities. Many of them related to DY's difficulties in assessing the risk that may be posed by others and her awareness of her own particular vulnerabilities. However, Knowles J considered that they were in fact more directed to an assessment of DY's capacity to make decisions about those with whom she had contact. "None of them supported a conclusion that DY lacked capacity to make decisions about relations generally or when she was distressed or unsettled" (paragraph 29).

Knowles J noted that, although the Court of Appeal had used the word "may" in relation to the information required by its reformulated test in *JB*, following submissions from the parties, she had "tempered her enthusiasm" to add to the list of relevant information. On this basis, and:

33. Standing back and considering the

issue in the round, I am satisfied that, at the time she was assessed, DY understood and was able to weigh all relevant information and had the capacity to decide to engage in sexual relations on a general non-specific basis. There was no dispute that she had the capacity to make decisions about sexual relations with her boyfriend, the only person with whom there was any current prospect of having sexual relations. The evidence of both Dr Camden Smith and YZ, DY's social worker, as to how she might respond in other circumstances was uncertain and speculative. She should not be assessed as lacking capacity unless all practicable steps have been taken to support her to make the decision without success, that included putting in place a package of support to limit and/or mitigate the effect of any periods of distress or unsettledness. The local authority's concerns about the risk of abuse and exploitation could be addressed through an appropriate package of care and contact arrangements, decided in DY's best interests. The prospective declaration proposed by the local authority was unworkable and imprecise.

The local authority had submitted that it would be "unconscionable if the court, recognising [DY's] vulnerability, held itself unable to protect her due to her situation not fitting within the framework of the MCA 2005." Whilst Knowles J understood the instincts prompting that submission, she noted that: "DY's care would be supported within the framework of the MCA 2005 as it is agreed by the parties, and I accept, that DY lacks capacity to make decisions about her care and contact with others. In those circumstances, the MCA 2005 provides an appropriate legal framework for a care package which protects DY from abuse and exploitation. I see no justification for invoking the inherent jurisdiction since it affords no greater scope for making a declaration of incapacity on grounds of a disturbance of the functioning of the mind than exists under the MCA 2005" (paragraph 34).

Knowles J therefore made a final declaration that DY **had** capacity to decide to engage in sexual relations, and directed the local authority "to prepare a care plan that will facilitate this in a way that reduces risk and supports DY to make informed decisions."

Comment

Starting at the end, it was on one view not open to Knowles J to direct the local authority to prepare a care plan to facilitate DY's ability to engage in sexual relations, not least in light of her finding that DY had capacity to make that decision. It is, however, entirely understandable why she sought to do so, not least to guide a local authority who were – again for understandable reasons – concerned as to navigate the line between protection and empowerment.

It might be thought on one view that DY's case could have been resolved much more simply on the basis of a person-specific test: i.e. by simply being allowed to ask whether DY had capacity to decide to engage in sexual relations with her boyfriend – to which the answer would clearly be yes (whether, in respect of any particular act between them, the act was consensual on both of their parts would be a different issue, falling to be considered, if at all, through the prism of the criminal law as would be the case with anyone). The question of whether DY had capacity to decide to engage in sexual relations would simply not fall to be considered by the court as it was not a live issue, and the court could properly decline to determine it (see, by analogy, <u>*Re SB*</u> [2020] EWCOP 43).

The case also clearly indicates the limits to the courts' increasing willingness to grapple with the concept of fluctuating capacity, as with (in a different context) the case of <u>CDM</u>, DY's situation could not sensibly be considered through this prism, and Knowles J was plainly correct to conclude that she had to determine either that DY had the capacity to make the material decision or she did not.

This case illustrates in clear terms the difficulties facing statutory bodies – local authorities and CCGs most obviously – tasked with safeguarding vulnerable individuals in the context of sexual relationships.

The jurisdiction of the Court of Protection is, as Knowles J's judgment reflects, a limited one; the inherent jurisdiction of the High Court, as the judgment also reflects, cannot fill the many gaps in protection the Court of Protection leaves unguarded. How to protect those who, like DY, fall into those gaps, is still unclear.

There is, in reality, some intellectual incoherence in the determination both that DY *has* capacity to make decisions on sex and the order that the local authority should make care plans to facilitate her sex life in a safe manner. Any lay person reading such a direction would rightly query: how can it be right that DY is assessed to be able to have sex but at the same time to require others to facilitate it? At the coal face of delivery, such orders can lead local authorities open to criticism for facilitating or encouraging sexual activity for vulnerable people in its care – see <u>Manchester City Council v LC</u> [2018] EWCOP 30; alternatively, they can face criticism for failing to protect the rights of those who wish to enjoy a sex life and have capacity to do so.

Analysis of the case law - compare the facilitative approach in TZ with the protective impulse in Manchester v LC; likewise JB in contrast to A Local Authority v B [2019] EWCA Civ 913 - does, arguably, reflect something of divergence of approach with regard to male and female service users. Perhaps this reflects the continued divergence of views with regard to male and female sexuality that society maintains. How this works in practice remains complex and difficult to advise upon - and may inevitably require a more and more tailored approach to individual capacity assessments as espoused by the Vice President in Tower Hamlets v NB and, in turn, more work for those tasked with caring for the vulnerable. How any of this works in the context of the s.27(1)(b) MCA 2005 exclusion of decisions regarding consent to sexual relations remains an open question.

PROPERTY AND AFFAIRS

Short note: permission to appeal refused in deputy charging case

In *Riddle v Public Guardian* [2021] EWCOP 38, Lieven J politely but firmly refused the deputy permission to appeal from this judgment, a very long and scrupulously careful decision of Senior Judge Hilder where she considered at considerable (necessary) length the ability of a professional deputy who is not a solicitor to charge fees at the solicitors' rate, concluding, broadly, that the deputy could not.

In coming to that conclusion, the Appeal Court usefully reminds us of the hurdles appellants face when trying to overturn a first instance judge's evaluative decision:

36. In my view this argument is bound to fail and doesn't come anywhere close to reaching the threshold for permission being granted to appeal. In deciding what weight should be given, the Judge needed to have some benchmark or comparator in undertaking the exercise. It was entirely reasonable for the Judge to look at a Local Authority and test against that, what rate would be reasonable in the circumstances. It seems to me to be significantly over forensic to argue that the Judge had to approach her analysis on what was the appropriate rate in any particular way, or that taking the comparison of the Local Authority was the wrong approach, or that she was starting in the wrong place. The judgment shows that the Judge had very carefully considered what services Mr Riddle was offering, and what rate was appropriate. I cannot see any arguable case on that point.

37. In truth, Mr Riddle's argument is a result of being in disagreement with the Judges' evaluative judgement and the weight she attached to various evaluative matters. Those are things for the first instance judge, and subject to any misdirection (of which there was none), this is not a matter that the appeal court should interfere with. It is further relevant in this particular regard to consider the experience of Her Honour Judge Hilder and her expertise in this field, and how she very carefully considered the facts of the particular cases before her.

Mencap publishes guides for the execution of LPAs

Mencap has just published a suite of "easy read" guides to the making and use of health and welfare LPAs, available <u>here</u>. They give easy to follow guidance on the setting up of such an LPA and then separate guidance to the attorney as to how the attorney should operate the LPA.

They are:

- The Supporter's Guide to the Health and Welfare Lasting Power of Attorney
- Easy Read Guide and Record for Making a Health and Welfare Lasting Power of Attorney
- Easy Read Guide for Sections 8 and 9 of a Health and Welfare Lasting Power of Attorney
- The Certificate Provider's Guidance: Health and Welfare Lasting Power of Attorney
- Guidance for a Health and Welfare Attorney

PRACTICE AND PROCEDURE

Consultation on hybrid and remote hearings

A rapid two-week consultation was launched on 10 June by the President of the Family Division to identify good practice from remote and hybrid hearings over the last 15 months to provide an evidence base to assist with decisions regarding of working future wavs as parties, representatives and judges return physically to This survey expressly includes court. consideration of the Court of Protection. The survey can be found here and the deadline for responses is 27th June 2021.

Remote hearings: the judicial view

In the early days of lockdown, then-Senior President of Tribunals, Sir Ernest Ryder, commissioned Dr Natalie Byrom and the Legal Education Foundation to carry out a survey of tribunals judicial office holders who had sat in remote hearings between 19 March 2020 and 31 July 2020, to gather feedback on their experiences. The report and the recommendations (dated June 2021) entitled 'Understanding the impact of COVID-19 on tribunals: The experience of tribunal judges' can be found here. Some of the experiences and challenges related will chime with experiences before the Court of Protection, hence the coverage in this section of the report.

The data was collected from judicial office holders via an online survey. It was open to all judicial office holders who had participated in hearings between 19 March 2020 and 31 July 2020. 1507 judicial office holders completed the survey in full including 670 from the Health, Education and Social Care Chamber.

The report considered a broad range of issues, from satisfaction with the technological aspects of remote hearings, through to the impact remote hearings has on access to justices, to the impact of remote hearings on judicial wellbeing. Of particular interest for our purposes are the parts of the report that consider the experiences of those participating in Mental Health Tribunals.

- Access to the hearings was a key issue to emerge. This was adversely affected by a lack of access to adequate equipment and broadband for those in detained settings, creating practical barriers to accessing remote hearings. Respondents reported patients being prevented from joining video hearings altogether, or forced them to attempt to participate from shared equipment. Further, respondents reported patients being too daunted by the thought of a remote hearing to attend at all.
- A further concern expressed was the impact on the hearing of the withdrawal of prehearing examinations of the patient by the medical member. The particular concerns articulated were:
 - The loss of the opportunity for the tribunal members to identify vulnerability in the absence of this interview, and the consequent lost opportunity to put in measures to address such vulnerability.
 - The lost opportunity to gain additional information on the patient's condition and general mental state.
 - The lost opportunity to establish rapport with the patient.

- A further theme to emerge was the impact on the patient of participating in a remote rather than an in person hearing. The issues raised in respect of this issue were multifactorial:
 - Concern were expressed that even if vulnerable patients were identified, the circumstances of the pandemic made it difficult to put in place effective adjustments that would enable patients to effectively participate. For example allowing them to sit with their lawyers, or with a supporter.
 - A number of respondents reported that proceeding with hearings by video or telephone risked exacerbating patient's symptoms, particularly where they were suffering from psychosis. It was reported by respondents that this had led to increased numbers of patients leaving their hearings part-way through or declining to attend in the first place.
 - A number of respondents reported experiences where patients had, unbeknownst to the panel, become very distressed during hearings and reacted with expressions of physical violence as a consequence. Other respondents reported concerns for the safety and wellbeing of nursing staff who were tasked with supporting patients to access remote hearings.
 - Respondents reported that remote hearings were longer than in person hearings, which in itself posed a barrier to communication with patients who were unable to cope with long hearings, and

left before they concluded.

The report makes a number of detailed recommendations. Of particular relevance to Mental Health Tribunals – and potentially the Court of Protection – are the following:

- Checks should be conducted to ensure that patients in detained mental health settings are able to access the technology and broadband required to participate in remote hearings. Hospitals should be asked to ensure that patients are able to join hearings from a quiet room and via a device that is not shared, and have access to writing materials to take notes.
- Leadership judges should issue guidance that remote hearings (both telephone and video) should be used with caution where:
 - Hearings are long;
 - Factual matters are in dispute;
 - Issues of credibility are engaged;
 - There are multiple participants and witnesses;
 - o Parties are unrepresented;
 - Parties require the assistance of an interpreter;
 - Parties experience physical or mental health difficulties, including hearing loss;
 - Issues are highly contested.

In practice, if these recommendations are followed, then once restrictions arising from the pandemic are lifted, remote hearings in mental health cases will not be common.

THE WIDER CONTEXT

CPR – Participation of Vulnerable Parties and Witnesses

The Civil Procedure Rules have been amended to include a new Rule (1.6) and Practice Direction 1A on the Participation of Vulnerable Parties or Witnesses. Factors which mav cause vulnerability in a party or witness include (but are not limited to): lack of understanding, communication difficulties, health condition, medical health condition or significant impairment of any aspect of their intelligence or social functioning (including learning difficulties). If the court decides that a party's or witness's availability to participate fully and/or give best evidence is likely to be diminished by reason of vulnerability, the court may order appropriate provisions to further the overriding objective, i.e. to ensure, so far as practicable, that the parties are on an equal footing and can participate fully in proceedings, and that parties and witnesses can give their best evidence. The court should also consider ordering "ground rules" before a vulnerable witness gives evidence, to determine what directions are necessary in relation to the nature and extent of that evidence, the conduct of the advocates and/or the parties, and/or any necessary support to be put in place for that person.

Short note: capacity in the rearview mirror

The immigration decision in [2021] UKAITUR HU135462019 is a very good example of how courts are getting themselves into a tangle thinking about capacity in retrospect. The immigration judge in the case had to consider whether the applicant in question had had capacity to marry. On appeal, Upper Tribunal Judge Allen noted that:

17. I consider first the judge's findings on the point of capacity. She set out at paragraph 64 the gist of section 1(2) of the Mental Capacity Act 2005, which provides that a person should be assumed to have capacity unless it is established that they do not. The judge went on to say that the difficulty in the case was that it was not established by any medical evidence before her upon which reliance could reasonably be placed that the sponsor did not have capacity in relation to a decision to marry. She went on to say that she shared the concerns raised by the respondent in relation to the quality of the capacity report but it did however provide a view of the social worker tasked with preparing such a report by her local authority employer that the sponsor did have capacity. The judge went on to say that she placed limited weight on that conclusion due to the concerns as to whether the assessment was properly and thoroughly focused on all relevant considerations, but the result remained a position where it was not established that the sponsor did not have capacity, and in such circumstances a person was to be assumed to have capacity. The judge regarded this as deeply unsatisfactory in all the circumstances of the case and given the evidence overall which she was shortly to go on to address as to the limitations on the sponsor's abilities, she had serious concerns as to the sponsor's capacity with respect to a decision to marry and thereafter engage in consequences including sexual relations. (emphasis added).

Whilst, for reasons irrelevant for present purposes, nothing ultimately turned on this

point, Upper Tribunal Judge Allen noted that:

As regards the findings on capacity I consider that the judge, though she wavered to an extent, clearly regarded herself as bound by section 1(2) of the 2005 Act and though she came close at times in her decision to going against the presumption, ultimately her decision was faithful to the presumption.

With respect, neither judge should have themselves bound considered by the presumption. The presumption operates in realtime, not in retrospect - and this case is a very good example of precisely why it should not do In circumstances where there was good SO reason to consider that the person did not have capacity to undertake the relevant act, it is selfevidently wrong that the presumption should operate to give the person the benefit of the doubt. The details for this are set out in more detail in this paper by Alex which was given at a webinar about the Clitheroe case concerning testamentary capacity, where Falk J appears to have laboured under the same misconception.

LeDer report published

The last Learning Disabilities Mortality Review report to be published by the University of Bristol is now <u>available</u>. The programme is continuing with some substantial changes (including the inclusion of autism), as detailed <u>here</u>.

The 2021 report focuses on findings from completed reviews of the deaths of people with learning disabilities that occurred in the calendar years 2018, 2019 and 2020, identifying any trends that have occurred over time. Because of the incremental roll out of the LeDeR programme in England during 2016 and 2017, 2018 was the first year in which the programme has relatively complete data. The report also includes analysis of the impact of COVID-19.

The overall summary of the Review is that:

There are some early indicators of improvements in the care of people with learning disabilities between 2018 and 2019, but there are also indications that such improvements are not felt across all aspects of service provision or groups of people with learning disabilities. Of particular concern are the significant inequalities in the experiences of people from minority ethnic groups. In addition, the COVID-19 pandemic has highlighted the impact of health inequalities and deficiencies in the provision of care of people with learning disabilities, with rates of their deaths being more than those of others'.

The inequalities and deficiencies are stark:

Compared to the general population, people with learning disabilities were more than 3 times as likely to die from an avoidable medical cause of death (671 per 100,000 compared to 221 per 100,000 in the general population). The majority of the excess was due to treatable medical causes of death.

Strengthening the Right to Independent Living

EHRC's briefing paper, <u>'Strengthening the Right</u> to Independent Living' published on 12 May 2021 discusses Article 19 of the UN Convention on the Rights of Persons with Disabilities (CRPD) and its impact in UK law:

This right to independent living is binding

under international law and the UK Government is expected to reflect its requirements in laws, policy and guidance. However, the right has not been fully incorporated into domestic law, meaning that disabled people have no redress in the UK courts if it is breached. In our view, the absence of effective legal protection for the right to independent living significantly limits disabled people's full and equal participation in society.

Across many areas of life, there is evidence that disabled people in Great Britain are not provided with the same choice, control and opportunities as others. Disabled people experience significant disparities in education and a persistent employment and pay gap. There is a chronic shortage of accessible housing and those with care needs often cannot access vital support. Thousands of disabled people are detained in institutions, out of sight and at risk of restrictive treatment or abuse. (pages 3-4)

The paper discusses the barriers faced by people with disabilities face in a number of different areas, and:

the action needed to ensure disabled people can enjoy the same freedoms, autonomy and opportunity as the rest of society (page 4).

The EHRC proposes that further measures should be taken:

to incorporate Article 19 rights and bring domestic law into line with international requirements. In collaboration with stakeholders, we have developed a legal model for incorporating the right to independent living into UK law. Our proposals have been endorsed by the UN Committee on the Rights of Persons with Disabilities as well as UK Parliament's Joint Committee on Human Rights (JCHR).

Our full proposed legal model is set out in the appendix to this paper. The key elements are:

- *a statutory Public Sector Inclusion Objective* that puts a duty on public bodies to act with the objective of meeting the requirements of CRPD Article 19
- a presumption on public bodies to provide accommodation, care and support in the community, unless this is not in line with the person's wishes
- recognition of the primacy of disabled people's views in decisions about accommodation, care and support, including a right to decline care
- a prohibition on the establishment of further institutional accommodation
- a duty on public bodies to assess the level of unmet need for accommodation in the community, and care and support to enable community or home living, and report on what they will do to meet that need, and
- effective enforcement mechanisms and provision of guidance on implementation to help ensure the right to independent living is upheld in practice. (page 8)

The report then goes on to set out in, depressing, detail, all the various ways in which Article 19 is not currently being upheld in England & Wales at present.

Seni's Law consultation

The Department of Health and Social Care has launched a consultation on draft statutory guidance on the Mental Health Units (Use of Force) Act 2018.

The Act has been passed, but is not yet in force. It is also as Seni's Law following the death of Olaseni Lewis at the age of 23 after being restrained at the Bethlem Royal Hospital, London in 2010 following levels of restraint described in the subsequent inquest by the coroner as 'disproportionate and unreasonable'.

The Act applies to both NHS and private mental health units (if at least some of the treatment is provided for the purposes of the NHS) and will require them to appoint a 'responsible person' to comply with the requirements of the Act. Mental health units will be required to have a published policy on the use of force (s.3) as well as providing information for patients (s. 4) and training for staff about the use of force (s. 5). Records will need to be kept of any but negligible use of force and the Secretary of State will come under a duty to prepare statistics and an annual report. It also requires police attending a mental health unit to take a body camera 'if reasonably practicable' (s. 12).

The proposed statutory guidance fleshes out some of the duties under the Act. Key points to note:

- As it is statutory guidance, there is a duty to have regard to it when exercising functions under the Act, and unless departures can be justified by a good reason they may give rise to legal challenges.
- 2. The Introduction notes that the use of force

appears to be increasing, and that the available data suggests there is often disproportionate use against certain groups of patients, including black and ethnic minority groups, women and girls, and people with autism and learning disabilities.

- 3. Guidance is given on what constitutes a use of force and restrictive interventions, and which settings the Act applies to.
- 4. The 'responsible person' does not need to be a new member of staff, but the appointment must be of a permanent member of staff, and someone of sufficient seniority i.e. at the level of Executive Director or equivalent.
- 5. The guidance sets out 16 points to be included as a minimum in policies on the use of force. These include details of what uses of force the organisation may use and in what circumstances, and how patients and their families will be involved in care planning. Before publication, the responsible person must consult with whoever they consider appropriate, which should include current and former patients and their families.
- In terms of information to be provided to patients, again the guidance sets out a checklist of what should be covered as a minimum, including the patient's rights and information about how to raise concerns. Again consultation is required before publication.
- 7. The draft guidance contains a substantial section on training, setting out the expectation that training should support an overall human rights based approach. It sets out areas which should be covered by training, and within those specific topics

which should be addressed;

- 8. In terms of recording, guidance is given on the exception for negligible use of force while also recording that it is already mandatory for NHS organisations or trusts and independent hospitals (where they are providing NHS-funded care), to submit data on the use of force to the NHS Digital Mental Health Services Data Set.
- The guidance on s.9 (investigations of deaths / serious incidents in mental health units whether or not force has been used) contains links to the existing NHS guidance on investigations, and a reminder of the NHS duty of candour.
- 10. Finally, guidance is given on delegation of the relevant person's responsibilities.

The draft statutory guidance can be found <u>here</u> and the link to the consultation is <u>here</u>. The consultation will run till midday 17 August 2021.

Readers interested in Seni's legacy may also be interested in the premiere of a new <u>documentary</u> concerning his death and mental health, injustice and art. In the summer of 2020, graffiti reading 'RIP SENI' was sprayed on an artwork made up of placards posing questions about mental capacity and assessment (ranging from 'are you free to determine your own actions?' to 'where are the resources and support for people who need them?'). The new film has been commissioned by the Bethlem Gallery and the Lewis family.

Research corner

We highlight here recent research articles of interest to practitioners. If you want your

article highlighted in a future edition, do please let us know – the only criterion is that it must be open access, both because many readers will not have access to material hidden behind paywalls, and on principle. This month we highlight two recent publications from the Mental Health and Justice Project appearing in Lancet Psychiatry, the first being an article: <u>Reasons for endorsing or rejecting selfbinding directives in bipolar disorder: a qualitative study of survey responses from UK service users</u>, and the second, a linked article including artworks from the artist embedded with the project team: <u>Self-binding directives</u> <u>through making - The Lancet Psychiatry</u>

Short note: voting, capacity and the ECHR on the CRPD

In Caamaño Valle v Spain [2021] ECHR 387, the European Court of Human Rights hardened yet further its stance that those with cognitive impairments can be disenfranchised if they do not have the mental capacity to vote, and did so in full knowledge that this contrary to the position adopted by the UN Committee on the Rights of Persons with Disabilities. It was notable for a very strong dissent from Judge Lemmens, who outlined how systems needed to be put in place to enable (through a surrogate if required) a voter with cognitive impairments to have their will and preferences reflected in a vote cast on their behalf. For a detailed discussion of the judgment, see the commentary on on Alex's website.

Fragmentation in international human rights law is a continuing, and serious, problem in the area of legal capacity. The ECtHR has shown itself quite willing to go toe-to-toe with the CRPD Committee in the context both of Article 5 (Rooman) and Article 8 ECHR (AM-V v Finland), in both instances finding that the CRPD Committee's interpretation of the CRPD did not mandate the radical change in the interpretation of the ECHR that some might have expected. It is clear from the strong dissent of Judge Lemmens that at least some within the ECtHR are concerned about what this means in terms of its role as guarantor of human rights. Conversely, and whilst the majority judgment in the instant case does seem almost willfully to be determined to stand in the face of the approach of the Committee, decisions such as AM-V v Finland can also be seen as legitimate demands from an experienced human rights court to the CRPD Committee to make out its case both that (1) its interpretation of the CRPD is, in fact, reflecting what the Convention requires; and (2) almost more importantly, that it is ethically right to make the leaps of faith that the Committee demand in a number of areas.

The position in the United Kingdom, it should be noted, is nuanced. There is no explicit prohibition on voting based upon mental incapacity (or a status such as mental disorder), s.73 Electoral Administration Act 2006 explicitly having abolished "[a]ny rule of the common law which provides that a person is subject to a legal incapacity to vote by reason of his mental state," but as Lucy Series explains in this blog, inadvertent barriers are placed in the way of individuals with cognitive impairments through requirements relating to registration. Moreover, in England & Wales s.29 MCA 2005 provides, expressly, that nothing in the MCA permits a decision on voting to be made on behalf of a person, although there are some interesting

questions which arise as to precisely how it can be tested that a person operating a proxy vote is doing so as the proxy for a person with capacity to vote (as the Electoral Commission envisages) or doing so on a surreptitious best interests basis. It would also be interesting to see whether and how policy makers would seek to enable decision-making on a 'will and preferences' basis within the UK context given the - obvious - concerns that must arise about the potential for the end result to reflect the views of the person constructing the will and preferences of the individual voter. However, given that the CRPD is not directly applicable in England & Wales, then unless the Grand Chamber in either the earlier Strobye and Rosenlund case or (if it goes there) this case finds that the ECHR should be interpreted in the fashion that Judge Lemmens sought, the difficulty of bringing any Human Rights Act 1998 challenge to s.29 MCA 2005 has only been increased. So it is not - yet - a question that policy makers within the English & Welsh context are likely to be required to answer by the courts (they may, interestingly, be required to in Scotland if the Scottish Government's stated intention to seek to incorporate the CRPD into Scottish law bears fruit).

Short note: the common law and the MCA – the view from Singapore

The English courts continue to grapple with the fact that the MCA 2005 does not set down a universal test of mental capacity or a universal decision-making framework for those with impaired capacity (see, most recently, *Re Clitheroe*; it is also an issue which will be likely to arise in *Re JB* in the Supreme Court). With thanks to Yue-En Chong for flagging this, the

June 2021 Page 46

Singapore Court of Appeal has weighed in on this point as well. This is of particular interest given the similarities between the Singapore MCA and the MCA 2005. In a complex contractual claim, one argument advanced was that

s. 19(1)(*c*) of the MCA [the equivalent of s.15(1)(*c*) MCA 2005, giving the power to the court to make a declaration of lawfulness] *empowers the court to annul* the second defendant's personal guarantee. In my view, the issue I have to resolve is as follows: does the Family Court's declaration that the second defendant lacked mental capacity oblige, or permit, annulling the Guarantee under s 19(1)(*c*)?

Dedar Singh Gill J held that:

100 Whether I am compelled to annul the Guarantee in light of the declaration of second defendant's the mental incapacity turns on whether the MCA is intended to override the common law requirement of proving the counterparty's knowledge of the mental incapacity. As our MCA is modelled after the UK's Mental Capacity Act 2005 (c 9) (UK) ("UK Mental Capacity Act") (Singapore Parliamentary Debates, Official Report (15 September 2008), vol 85 at col 109 (Dr Vivian Balakrishnan, then Minister for Community Development, Youth and Sports)), materials which elucidate the legislative intent and scope of the UK Mental Capacity Act are instructive for our purposes. It is clear to me that the UK Mental Capacity Act is not intended to displace the rule at common law that, in general, a contract entered into by a person who lacks capacity to contract is voidable only if the other contracting party has actual or constructive knowledge of the lack of capacity (Explanatory Notes to the UK Mental Capacity Act at [45]). Consequently, a declaration of mental capacity under our MCA does not by itself annul the incapacitated party's contract, especially one concluded prior to the declaration.

101 Notwithstanding the above, does the court have a residual discretion to annul a contract concluded by a mentally incapacitated person under s 19(1)(c) of the MCA where the contract was concluded prior to the declaration of mental incapacity? Even if I accept that the court has such a discretion, the second defendant has not furnished the grounds on which I should exercise it in his favour. The Court of Appeal decision cited by the second defendant, Re BKR, concerns the setting aside of a trust and a transfer of assets from two banks to a third which were created and/or effected prior to the declaration of the third respondent's mental incapacity. However, Re BKR does not stand for the proposition that the courts should similarly intervene in contractual relations.

102 In conclusion, I am not prepared to annul the Guarantee under s 19(1)(c) MCA on the basis of the second defendant's mental incapacity where the common law does not see fit to do so.

SCOTLAND

The hospital discharge scandal: MWC Report

Mental Welfare Commission for Scotland ("MWC") has issued its Report "Authority to discharge: Report into decision making for people in hospital who lack capacity". While a few comments and criticisms of the Report are Commission appropriate, the deserves congratulations for a huge amount of investigatory research followed by careful and thoughtful analysis, carried out - having regard to the sheer size of the task - within a commendably tight timescale; notwithstanding that the Report itself makes for depressing, indeed alarming, reading, with evidence-based confirmation of the worst fears already arising from the various matters that we reported and commented upon in the October and November Reports.

The Report is probably the most significant work of the Commission so far under the leadership of Julie Paterson, Chief Executive, who introduced the Report as follows:

"People who lack mental capacity and who are being cared for and treated in care homes and hospitals are among the most vulnerable in our society. The focus of this report was to examine the detail of a sample number of hospital to care home moves of people from across Scotland, to check that those moves were done in accordance with the law during the early stages of the pandemic.

"Some of our concerns relate specifically to the significant pressures of the pandemic.

But, worryingly, the report also finds more endemic examples of poor practice. Lack of understanding of the law, lack of understanding of good practice, confusion over the nature of placements, misunderstanding over power of attorney. These findings are very disappointing and may mean that many more moves were made without valid legal authority.

"This report also finds a lack of uniformity from one HSCP to another, with different approaches to national legislation and guidance adopted in different areas.

"We call on Health and Social Care Partnerships across Scotland, the Care Inspectorate and the Scottish Government to read our report in detail and act swiftly on our recommendations".

The starting-point for the Commission's work was the same cohort of 5,204 patients discharged from NHS hospitals to care homes in Scotland from 1st March to 31st May 2020 on whom we commented previously. Political controversy now engulfs the Public Health Scotland ("PHS") Report after it was widely drawn to public attention that part of the remit of PHS is to safeguard the reputations of Scottish Ministers. For purposes of commentary from the viewpoint of law and human rights, that political controversy is largely irrelevant, as the PHS Report was totally damning by omission, in that it did not address, nor did it disclose that anyone addressed, the need for legality, or the concept that the patients affected were not in fact inanimate and inconvenient blockages occupying hospital beds (which is how the PHS Report seems to portray them as having been

treated), but individual human beings with the same rights, including fundamental human rights, as everyone else.

MWC set out with the objective of investigating a sample of 10% of those 5,204 discharges. MWC sought information from Health and Social Care Partnerships ("HSCPs") in relation to people moved from hospital to registered care home settings during that three-month period. HSCPs were described as "very responsive" to that request, with the exception of Highland, an exception perhaps worthy of further attention. From those returns MWC sought information about 731 of those people, 465 of whom were reported by HSCPs to have lacked capacity to agree to a move from hospital to a care home. Eight of those did not fulfil the inclusion criteria for the enquiry, so the work focused on 457 people. Thus, the MWC Report requires to be considered from the starting-point that all of those investigated were unable to give valid consent to the move. Headlines suggesting that 20 of those were unlawfully moved are misleading. For 20 of them it appears that no potential cloak of legality was even offered. If one moves the focus from whether there was any such cloak of legality and how thin that cloak might have been in individual cases, which would not unfairly summarise MWC's approach, to asking how many of those discharges failed demonstrably to comply with the requirements for lawfulness in terms of Article 5 of the European Convention on Human Rights, as interpreted in relevant jurisprudence, the answers are different. This article uses "lawful" and "unlawful" by reference to that Article 5 test.

For a start, 90 moves (20% of the total) were claimed to have been authorised by section

13ZA of the Social Work (Scotland) Act 1968. However, those 90 all lacked relevant capacity, and therefore required a lawful process that was compliant with Article 5 in order to render those discharges lawful, even though - absurdly some of those in the total investigated were contradictorily described by relevant HSCPs as being both incapable of consenting yet having consented. The MWC Report refers to the 2007 guidance to local authorities on their powers under the 1968 Act, issued by Scottish Executive, but does not appear to narrate that that guidance is clear that section 13ZA should not be used where the proposed move would result in an Article 5 deprivation of liberty. It also appears to fail to mention that the reason for this was simple: section 13ZA does not provide for a procedure that would comply with Article 5. Local authorities are bound to comply with such guidance by virtue of section 5 of the 1968 Act. It would thus appear that all of those 90 discharges were unlawful.

267 of the moves were reported to have been authorised under powers of attorney. The relevant section of the MWC Report is a litany of failures to check almost all of the basic necessary requirements before it could be accepted that the person claiming to be the attorney was appropriately empowered. In 78 cases it was admitted that the power of attorney document had not even been read, and in a further 61 cases no details had been recorded. and indeed in an undisclosed proportion of those discharge had proceeded on the basis of being "advised of the contents" rather than reading them. In some cases the attorney held only financial powers, or the document had not been registered with the Office of the Public Guardian. In 70 cases it was either unknown or not

recorded whether the power of attorney authorised the move. In 77 cases it was "unclear if the powers had been validly triggered". Three significant further points do not appear to be addressed in the Report.

Firstly, it is not reported whether in any of these 267 cases there was any check with the Office of the Public Guardian as to whether – even where a certified power of attorney document had been produced – the document remained in force, with neither the document itself nor any of the powers in it revoked.

Secondly, the MWC Report does not appear to address the question of whether the power of attorney documents empowered the attorney purportedly to authorise a deprivation of liberty, regardless of whether it could be said to "authorise the move". Some lawyers dispute whether a power of attorney can ever confer power to authorise a deprivation of liberty, and although that point has been raised in litigation it does not appear ever to have been authoritatively determined. The less restrictive view is that such power can be conferred, but having regard to decisions such as McDowall's Executors v IRC [2004] STC (SCD) 22 - the document must clearly and explicitly empower the attorney to authorise a deprivation of liberty.

Thirdly, exercise of such power is only one preliminary element for Article 5 compliance. For example, it must be shown that the person in question has a "real and effective" ability to test the lawfulness of the deprivation of liberty. The Strasbourg Court has made clear that this must not be an illusory and theoretical right, but one which practically and actively assists in the appeal process. See also the "checklist" referred to at the end of the third item in this section of this Report.

One cannot deduce from the MWC Report that it was clearly and fully established that any of those 267 moves was lawful.

Similar doubts and reservations attach to many of the moves said to have been authorised by a welfare guardian. In some cases, a guardianship order had been applied for, but had not yet been granted, and there had been no interim order. The terms of the guardianship orders had not been checked and recorded. There was no record of any check with the Office of the Public Guardian as to whether the order produced was still in force, and that neither the order nor any of the powers conferred had been revoked.

One can only have sympathy with the pressures under which those trying to manage these discharges were placed. In a telling passage, MWC writes that:

"It was also suggested that those involved in discharge planning were under significant pressure to manage delayed discharges, which felt like a process of "emptying beds" and it was a "battle" to retain focus on the person. Whilst this was exemplified by the pandemic, it was explained that the pressures relating to delayed discharge processes have been long standing and challenging. Without understanding of what may constitute a deprivation of liberty, practice may well be flawed, with consequent impact on the rights of the individual who lacks capacity. Discharges from hospital to care homes bring this into sharp focus and practitioners require high levels of training, support and leadership to fulfil their functions to ensure that any moves are lawful and compliant with an individual's human rights, as well as their economic, social and cultural rights."

That echoes our assessments in relevant items in the October and November Reports.

The background of unlawfulness and endemic and widespread ageism and disability discrimination that we have narrated in various relevant Reports dates back well before the pandemic, and continues. In December 2018, MWC first publicly reported a policy of NHS Greater Glasgow and Clyde moving patients from hospital into various care homes without obtaining either valid consent of the patient or, where they were unable to give that, relevant legal authority. The patients were detained in the care homes and were prevented from leaving. These were clear violations of Article 5. Then a lady held against her will for over a year applied to a Mental Health Tribunal for her release. The units weren't hospitals, therefore the Tribunal did not have jurisdiction, but expressed concern. EHRC commenced the EHRC action, initially defended. Then on 20th November 2020 the Commission issued a statement "Equality and Human Rights Commission reaches settlement on ending unlawful detention of adults with incapacity by NHS Great Glasgow and Clyde". NHS Greater Glasgow and Clyde, and the owners of the chain of care homes, acknowledged that their practice was without legal authority and was unlawful.

We have had the PHS Report and the comment that it has attracted, and now the MWC Report, yet widespread unlawfulness continues. Latest reports are that hospital patients who might previously have been unlawfully decanted into care homes are now simply being left on medical wards. It is reported that all wish to return home, all are medically fit for discharge, but some have left and been returned by the police. Apart from these attempted escapes, none have been off the ward in several months even for a walk. They literally sit in a medical ward on a bed all day. This is an unlawful situation. The Supreme Court has made it amply clear that it is a deprivation of liberty to keep a patient in hospital, except with valid informed consent of the patient, beyond the point where life-saving treatment is immediately necessary and where the person is clinically fit for discharge. Overall, the MWC Report should be read in full by all concerned with aspects of the management of hospital patients at and beyond the point where life-saving treatment is no longer immediately necessary and the person is clinically fit for discharge, whether actually discharged or not. It is inconceivable that this impressive Report will not at last result in substantial improvements in this continuing situation, or that there should be any avoidable delay in implementing in full the eight recommendations addressed to HSCPs, the two addressed to the Care Inspectorate, and the final one addressed to Scottish Government.

A final requirement not narrated in the MWC Report is at long last to put in place, with the minimum of delay, appropriate provision for authorising deprivations of liberty, at a very minimum those arising when hospital patients no longer require to be there for life-saving treatment that is immediately necessary, and are clinically fit for discharge. On a possible continuation of that theme, see the next item.

Adrian D Ward

Change at Scottish Government – but how much?

We looked forward to what might be achieved in the first year of the new Parliament in the February Report and to the elections themselves in March. It was known that Jeanne Freeman. former Cabinet Secretary for Health and Social Care, was standing down. It must be positive news that Humza Yousaf has had his abilities and experience transferred from the Justice brief to become Cabinet Secretary for Health and Social Care. Since before the establishment of the Scottish Parliament, territorialism as between justice and health interests has from time to time presented difficulties that have required conscious efforts to bridge. Thus the Bill for the Adults with Incapacity (Scotland) Act 2000 was dealt with principally by the Justice and Home Affairs Committee, whereas the Bill for the Adult Support and Protection (Scotland) Act 2007, which included the last round of previous significant reform to the 2000 Act, was allocated to the Health Committee, though at that stage it certainly assisted continuity that Roseanna Cunningham MSP was convener of each committee at the relevant time. There have nevertheless been divergences, particularly when aspects of Part 5 of the 2000 Act have been directed and viewed as a separate island with inadequate respect for its strong links to the rest of the Act.

Kevin Stewart has been appointed Minister for Mental Wellbeing and Social Care, effectively replacing Clare Haughey, who has become Minister for Children and Young People. It is perhaps understandable that as a former mental health nurse, mental health issues remained closest to her heart. The announcement of the establishment of the Scott Review on 19th March 2019 placed substantial emphasis on mental health aspects. It did contain the following paragraph:

"At the same time as the review takes place, we will complete the work we have started on reforms to guardianships, including work on restrictions to a person's liberty, creation of a short term placement and amendments to power of attorney legislation so that these are ready when the review is complete."

Nevertheless, AWI reform, even the seriously urgent need for deprivation of liberty provision, effectively stalled for two years. There are however now positive signs that at least as regards deprivation of liberty issues when patients are ready for discharge from hospital, the pace may now pick up again. In the January Report we already referred to the possibility of legislation in the first session of the new Parliament, and in the February Report we referred to establishment of the "AWI Emergency Commencement Consideration Legislation Group", which became so concerned about the topic of unlawful discharges that it has morphed into a group advising on the prospect of at least some early AWI reform. We shall report any decisions by the new ministerial team to move forward with this. If so, one may expect a period of public consultation on whatever might be proposed.

Biographies of the Scottish Cabinet and Ministers are at: <u>https://www.gov.scot/about/who-runs-</u> <u>government/cabinet-and-ministers/</u>.

Adrian D Ward

Welfare guardianship and deprivation of liberty

The decision of the Sheriff Appeal Court in <u>JK</u> (Respondent & Appellant) v Argyll and Bute Council ("the Council") (Applicant & Respondent), a decision bearing the reference DNN-AW15-19, helpfully addresses the fundamental question as to whether under the Adults with Incapacity (Scotland) Act 2000 ("the 2000 Act") powers can be granted to a welfare guardian which have the effect of depriving the adult of his or her liberty, and also the inter-relationship between sections 64 and 70 of the 2000 Act. The decision extends. in effect, to a helpful refresher course on the purpose, approach and principal relevant provisions of the 2000 Act. **Rather** disappointingly, the decision lacks a similarly helpful analysis of the steps required under Article 5 to render lawful such a deprivation of liberty as was contemplated in the case.

At first instance the sheriff concluded that JKwas suffering from a serious delusional disorder substantially affecting her everyday life, and resulting in levels of self-neglect and lack of personal hygiene that were potentially lifethreatening. She required and would continue to require constant support with all aspects of her life because she lacks the cognitive ability to look after her own basic needs. The sheriff held that the adult was incapable in relation to decisions about, or acting to safeguard, her personal welfare and was likely to continue to be so incapable for the foreseeable future. He demonstrated that the section 1 principles would be met by granting a guardianship order. Having considered a proposed care package for the adult, the sheriff took the view that many of the powers craved were not necessary. He concluded that it was essential to grant the following powers:

"(a) The power to decide where the adult should live, to require her to live at that location, to convey her to that location and to return her to that location in the event of her absenting herself therefrom

"(j) The power to decide where the adult is permitted to go and decide whether or not the adult should be accompanied by a person nominated by her guardian to assist with her personal safety and welfare."

Those two powers in particular were, as the Appeal Court put it, bitterly contested. In practical terms, the issue centred upon whether the adult should remain in her own home, despite the disadvantages and risks of doing so, as she wished, or be transferred with the care package to a secure facility where she would be deprived of her liberty, but within that constraint afforded suitable accommodation and care, and as much freedom as could reasonably be allowed.

The sheriff granted powers (a) and (j) but limited the duration of the order to one year for the express purpose of permitting the court to review progress with rehabilitation, and the deprivation of liberty, within a reasonable timeframe.

The principal argument for *JK* in the appeal was that the sheriff could not competently grant to a welfare guardian powers which would have the effect of depriving the adult of liberty unless the 2000 Act contained express authority to that effect, which it did not. The argument for *JK* founded upon *Welsh Ministers* v *PJ* [2020] AC 757 at paragraphs 24 and 25, and MM v Secretary of State for Justice [2019] AC 712 at paragraph 31. For the Council it was submitted that the 2000 Act does afford the power sought, and that under reference to Stanev v Bulgaria (2012) 55 EHRR 22 the 2000 Act contains a carefully constructed system which is Article 5 compliant, which allows the adult to express the adult's views, past and present, and under which measures can only be authorised where they are necessary. The Appeal Court accepted that powers (a) and (j) were contrary to the expressed wishes of the adult, and amount to a deprivation of her liberty and are an interference with her right to choose her place of residence. However, they were granted in the specific context of safeguarding the welfare of an adult who lacked relevant capacity and whose living conditions, self-neglect and lack of personal hygiene were potentially life-threatening. The Appeal Court held that the grant of the powers sought accorded with the existing line of sheriff court decisions (Muldoon, Applicant 2005 SLT (Sh Ct) 52, M, Applicant 2009 SLT (Sh Ct) 185, and Scottish Borders Council v AB 2020 SLT (Sh Ct) 41). The Appeal Court pointed out that section 64 is in wide terms. It does not contain a lengthy list of possible powers. It includes power to deal with all ("all" emphasised by the court) aspects of the personal welfare of the adult, as well as power to deal with particular matters. Any powers had to be granted consistently with the section 1 principles. While section 64 does not contain an explicit power to detain, it can subject to the application of the principles - be deployed to address situations such as that of JK. The court was satisfied that given the factual matrix, "a matter" or "all aspects" in section 64 would cover the welfare issues of transitioning an adult from one form of care and accommodation to another form of care and facility, ensuring that the adult remains in the facility to address the adult's needs, and returning the adult there should she leave. The Appeal Court derived no assistance from the two cases cited for *JK* as they "involve an entirely different scheme in a different factual matrix".

On the relationship between sections 64 and 70, the Appeal Court held – in short – that it could not be argued that section 70 detracted from the power to authorise a deprivation of liberty under section 64, because section 70 contained additional safeguards to enforce a decision of a guardian in accordance with welfare powers when it is not complied with, therefore section 70 and its safeguards cannot be engaged unless the relevant powers have been granted under section 64 in the first place.

A ground of appeal suggesting that the order would remove from the adult any right to litigate against the Council "was not pressed ... to any material extent", but in any event the Appeal Court held that the application did not seek any power on the part of the welfare guardian to make decisions for the adult in relation to any litigation which might be brought by the adult against the local authority. The Appeal Court held that the adult was not constrained in that regard by the provision of section 64(3) that a guardian has power to act as the adult's representative in relation to any matter within the scope of the powers conferred by the guardianship order.

Surprisingly, as indicated above, the Appeal Court did not go through the actual steps required for compliance with Article 5. Thus, for example, the Appeal Court did not address whether or not it had been proved that the adult was "of unsound mind", a prerequisite for the engagement of Article 5(1)(e). One would also have preferred the Appeal Court to "tick off" the other requirements for Article 5 compliance, expressly rather than by implication. A useful guide to these requirements can be found in <u>Appendix B</u> to the Report of the Law Commission of England & Wales on Mental Capacity and Deprivation of Liberty.

Habitual residence of a child

Always subject to the major caveat that "adults with impaired capacity are not big children", decisions relating to children can sometimes, with due caution, be of assistance in cases concerning adults. <u>K v D</u> was a decision by Sheriff A M Mackie in Glasgow Sheriff Court on 19th February 2021. Proceedings had commenced in Glasgow Sheriff Court in which interim residence and contact orders had been made with the parties' agreement in November 2015. Mother then moved from Glasgow to Liverpool without telling father or obtaining his consent or a court order permitting the relocation. Mother then sought dismissal of the Scottish proceedings in their entirety, submitting that the English courts had acquired exclusive jurisdiction on the basis that the child's habitual residence had changed from Glasgow to Liverpool, and that the Glasgow court was forum non conveniens. Mother's motion was refused. It was undisputed that the Glasgow court had had jurisdiction when the proceedings were commenced. The Glasgow court was not satisfied that there was another court of competent jurisdiction enabled to consider and rule on the residence and contact arrangements. The child's habitual residence had not been changed by the unilateral action of the pursuer.

June 2021 Page 54

Adrian D Ward

Evidence of solicitor for granter of Will

Thompson v Hopkinson [2021] SAC (Civ) 14, decided on 22nd March 2021, was yet another case in which a disappointed relative of a person not benefiting from the Will of a relative, sought production and reduction of the Will on grounds that it was impetrated by facility and circumvention. A particular point at issue was whether evidence should be admitted from the solicitor who prepared the Will and arranged for it to be executed, on the grounds that the pleadings gave no fair notice and lacked candour. It was held that that ground of appeal was misconceived and arose from а misunderstanding of the nature and function of written pleadings. Both parties had been "economical in their pleadings". The appellant's pleadings set out the deceased's physical ailments and amounted to bold assertions and bare statements without any specification, beyond a reference to confusion on the part of the deceased. It would have been sufficient for the respondent to have met those averments with a simple denial. In fact, the respondent averred that the deceased was not facile. That was a positive statement which the respondent had offered to prove. It was sufficient. It gave fair notice that the evidence to be elicited from the solicitor would have included evidence addressing the deceased's state of mind.

Moreover, for the appellant to have succeeded he would have required to establish not only the facility of the deceased at the time when the Will was made, but acts of circumvention or fraud, and lesion. The appellant's evidence of facility was inadequate and was directly contradicted by the solicitor's evidence. There was scant evidence of lesion. There was insufficient circumstantial evidence before the sheriff from which circumvention could be inferred.

Adrian D Ward

Standing of ECHR Article 6

A criminal case not directly relevant to adult incapacity matters nevertheless contained the significant observation that ECHR, and particularly Article 6, did not form a separate code which was applicable, independent of domestic principles of fairness, but rather that it permeated the whole system. The case was *Darbazi v HM Advocate* [2021] HCJAC 10.

Adrian D Ward

Electronic and digital Wills

The regimes for execution of Wills, and of powers of attorney, are distinct, but discussion of either regime often takes account, for comparative purposes, of the other. Articles by John Kerrigan are always worth reading. Practitioners in adult incapacity law may therefore find interesting his article on "Electronic and digital Wills" at 2021 SLT (News) 25.

Adrian D Ward

Fditors and Contributors

Alex Ruck Keene: alex.ruckkeene@39essex.com

Alex is recommended as a 'star junior' in Chambers & Partners for his Court of Protection work. He has been in cases involving the MCA 2005 at all levels up to and including the Supreme Court and the European Court of Human Rights. He also writes extensively, has numerous academic affiliations, including as Visiting Professor at King's College London, and created the website www.mentalcapacitylawandpolicy.org.uk. To view full CV click here.

Victoria Butler-Cole QC: vb@39essex.com

Victoria regularly appears in the Court of Protection, instructed by the Official Solicitor, family members, and statutory bodies, in welfare, financial and medical cases. Together with Alex, she co-edits the Court of Protection Law Reports for Jordans. She is a contributor to 'Assessment of Mental Capacity' (Law Society/BMA), and a contributor to Heywood and Massey Court of Protection Practice (Sweet and Maxwell). To view full CV click here.

Neil Allen: neil.allen@39essex.com

Neil has particular interests in ECHR/CRPD human rights, mental health and incapacity law and mainly practises in the Court of Protection and Upper Tribunal. Also a Senior Lecturer at Manchester University and Clinical Lead of its Legal Advice Centre, he teaches students in these fields, and trains health, social care and legal professionals. When time permits, Neil publishes in academic books and journals and created the website www.lpslaw.co.uk. To view full CV click here.

Annabel Lee: annabel.lee@39essex.com

Annabel has experience in a wide range of issues before the Court of Protection, including medical treatment, deprivation of liberty, residence, care contact, welfare, property and financial affairs, and has particular expertise in complex cross-border jurisdiction matters. She is a contributing editor to 'Court of Protection Practice' and an editor of the Court of Protection Law Reports. To view full CV click here.

Nicola Kohn: nicola.kohn@39essex.com

Nicola appears regularly in the Court of Protection in health and welfare matters. She is frequently instructed by the Official Solicitor as well as by local authorities, CCGs and care homes. She is a contributor to the 5th edition of the Assessment of Mental Capacity: A Practical Guide for Doctors and Lawyers (BMA/Law Society 2019). To view full CV click here.

Katie Scott: katie.scott@39essex.com

Katie advises and represents clients in all things health related, from personal injury and clinical negligence, to community care, mental health and healthcare regulation. The main focus of her practice however is in the Court of Protection where she has a particular interest in the health and welfare of incapacitated adults. She is also a qualified mediator, mediating legal and community disputes. To view full CV click here.

Rachel Sullivan: rachel.sullivan@39essex.com

Rachel has a broad public law and Court of Protection practice, with a particular interest in the fields of health and human rights law. She appears regularly in the Court of Protection and is instructed by the Official Solicitor, NHS bodies, local authorities and families. To view full CV click here.















Stephanie David: stephanie.david@39essex.com

Steph regularly appears in the Court of Protection in health and welfare matters. She has acted for individual family members, the Official Solicitor, Clinical Commissioning Groups and local authorities. She has a broad practice in public and private law, with a particular interest in health and human rights issues. She appeared in the Supreme Court in *PJ v Welsh Ministers* [2019] 2 WLR 82 as to whether the power to impose conditions on a CTO can include a deprivation of liberty. To view full CV click <u>here</u>.

Arianna Kelly: arianna.kelly@39essex.com

Arianna has a specialist practice in mental capacity, community care, mental health law and inquests. Arianna acts in a range Court of Protection matters including welfare, property and affairs, serious medical treatment and in matters relating to the inherent jurisdiction of the High Court. Arianna works extensively in the field of community care. To view a full CV, click <u>here</u>.

Simon Edwards: simon.edwards@39essex.com

Simon has wide experience of private client work raising capacity issues, including *Day v Harris* & *Ors* [2013] 3 WLR 1560, centred on the question whether Sir Malcolm Arnold had given manuscripts of his compositions to his children when in a desperate state or later when he was a patient of the Court of Protection. He has also acted in many cases where deputies or attorneys have misused P's assets. To view full CV click <u>here</u>.

Scotland editors

Adrian Ward: adw@tcyoung.co.uk

Adrian is a recognised national and international expert in adult incapacity law. He has been continuously involved in law reform processes. His books include the current standard Scottish texts on the subject. His awards include an MBE for services to the mentally handicapped in Scotland; honorary membership of the Law Society of Scotland; national awards for legal journalism, legal charitable work and legal scholarship; and the lifetime achievement award at the 2014 Scottish Legal Awards.

Jill Stavert: j.stavert@napier.ac.uk

Jill Stavert is Professor of Law, Director of the Centre for Mental Health and Capacity Law and Director of Research, The Business School, Edinburgh Napier University. Jill is also a member of the Law Society for Scotland's Mental Health and Disability Sub-Committee. She has undertaken work for the Mental Welfare Commission for Scotland (including its 2015 updated guidance on Deprivation of Liberty). To view full CV click <u>here</u>.













June 2021



Conferences

Members of the Court of Protection team are regularly presenting at webinars arranged both by Chambers and by others.

Alex is also doing a regular series of 'shedinars,' including capacity fundamentals and 'in conversation with' those who can bring light to bear upon capacity in practice. They can be found on his <u>website</u>.

Neil is doing a DoLS refresher (by Zoom) on 29 June 2021. For details and to book, see <u>here</u>.

Neil and Alex are doing a joint DoLS masterclass for mental health assessors (by Zoom) on 12 July 2021. For details, and to book, see <u>here</u>.

Advertising conferences and training events

If you would like your conference or training event to be included in this section in a subsequent issue, please contact one of the editors. Save for those conferences or training events that are run by non-profit bodies, we would invite a donation of £200 to be made to the dementia charity My Life Films in return for postings for English and Welsh events. For Scottish events, we are inviting donations to Alzheimer Scotland Action on Dementia.

Our next edition will be out in July. Please email us with any judgments or other news items which you think should be included. If you do not wish to receive this Report in the future please contact: marketing@39essex.com.

Sheraton Doyle Senior Practice Manager sheraton.doyle@39essex.com

Peter Campbell Senior Practice Manager peter.campbell@39essex.com



Chambers UK Bar Court of Protection: Health & Welfare Leading Set



The Legal 500 UK Court of Protection and Community Care Top Tier Set

clerks@39essex.com • DX: London/Chancery Lane 298 • 39essex.com

LONDON

81 Chancery Lane, London WC2A 1DD Tel: +44 (0)20 7832 1111

MANCHESTER

82 King Street, Manchester M2 4WQ Tel: +44 (0)16 1870 0333 Eax: +44 (0)20 7353 3978 59 Essex Chambers is an equal opportunities employer.

SINGAPORE

Maxwell Chambers, #02-16 32, Maxwell Road Singapore 069115 Tel: +(65) 6634 1336

KUALA LUMPUR

#02-9, Bangunan Sulaiman, Jalan Sultan Hishamuddin 50000 Kuala Lumpur, Malaysia: +(60)32 271 1085

39 Essex Chambers LLP is a governance and holding entity and a limited liability partnership registered in England and Wales (registered number 0C360005) with its registered office at 81 Chancery Lane, London WC2A 1DD.

39 Essex Chambers' members provide legal and advocacy services as independent, self-employed barristers and no entity connected with 39 Essex Chambers provides any legal services. 39 Essex Chambers (Services) Limited manages the administrative, operational and support functions of Chambers and is a company incorporated in England and Wales (company number 7385894) with its registered office at 81 Chancery Lane, London WC2A 1DD.